# Eastern Illinois University The Keep

**Masters Theses** 

**Student Theses & Publications** 

1-1-2012

# African American students' perceptions of undergraduate experiences: Matriculation, retention, attrition, and graduation

Christina L. Rainer *Eastern Illinois University* 

This research is a product of the graduate program in Counseling and Student Development at Eastern Illinois University. Find out more about the program.

#### Recommended Citation

Rainer, Christina L., "African American students' perceptions of undergraduate experiences: Matriculation, retention, attrition, and graduation" (2012).  $Masters\ Theses.\ 790.$  http://thekeep.eiu.edu/theses/790

This Thesis is brought to you for free and open access by the Student Theses & Publications at The Keep. It has been accepted for inclusion in Masters Theses by an authorized administrator of The Keep. For more information, please contact tabruns@eiu.edu.

# \*\*\*\*\*\*US Copyright Notice\*\*\*\*\*

No further reproduction or distribution of this copy is permitted by electronic transmission or any other means.

The user should review the copyright notice on the following scanned image(s) contained in the original work from which this electronic copy was made.

Section 108: United States Copyright Law
The copyright law of the United States [Title 17,
United States Code] governs the making of
photocopies or other reproductions of copyrighted
materials.

Under certain conditions specified in the law, libraries and archives are authorized to furnish a photocopy or other reproduction. One of these specified conditions is that the reproduction is not to be used for any purpose other than private study, scholarship, or research. If a user makes a request for, or later uses, a photocopy or reproduction for purposes in excess of "fair use," that use may be liable for copyright infringement.

This institution reserves the right to refuse to accept a copying order if, in its judgment, fulfillment of the order would involve violation of copyright law. No further reproduction and distribution of this copy is permitted by transmission or any other means.

#### THESIS MAINTENANCE AND REPRODUCTION CERTIFICATE

TO: Graduate Degree Candidates (who have written formal theses)

SUBJECT: Permission to Reproduce Theses

An important part of Booth Library at Eastern Illinois University's ongoing mission is to preserve and provide access to works of scholarship. In order to further this goal, Booth Library makes all theses produced at Eastern Illinois University available for personal study, research, and other not-for-profit educational purposes. Under 17 U.S.C. § 108, the library may reproduce and distribute a copy without infringing on copyright; however, professional courtesy dictates that permission be requested from the author before doing so.

#### By signing this form:

- · You confirm your authorship of the thesis.
- You retain the copyright and intellectual property rights associated with the original research, creative activity, and intellectual or artistic content
  of the thesis.
- You certify your compliance with federal copyright law (Title 17 of the U.S. Code) and your right to authorize reproduction and distribution of all
  copyrighted material included in your thesis.
- You grant Booth Library the non-exclusive, perpetual right to make copies of your thesis, freely and publicly available without restriction, by
  means of any current or successive technology, including but not limited to photocopying, microfilm, digitization, or Internet.
- You acknowledge that by depositing your thesis with Booth Library, your work is available for viewing by the public and may be borrowed through
  the library's circulation and interlibrary departmenst or accessed electronically.
- You waive the confidentiality provisions of the Family Educational Rights and Privacy Act (FERPA) (20 U.S.C. § 1232g; 34 CFR Part 99) with
  respect to the contents of the thesis, including your name and status as a student at Eastern Illinois Univerity.

#### Petition to Delay:

I respectfully petition that Booth Library delay maintenance and reproduction of my thesis until the date specified and for the reasons below. I understand that my degree will not be conferred until the thesis is available for maintenance and reproduction.

Date: 5 23 12

Reasons:

Author's Signature

Date

ist lainer 5/23/12

This form must be submitted in duplicate.

Matriculation. Retention	n, Attrition, and Graduation	
	(TITLE)	-
	BY Poince	
Christi	na L. Rainer	
•	THESIS	
	LFILLMENT OF THE REQUIREMENTS HE DEGREE OF	
Master of Science i	in College Student Affairs	
IN THE GRADUATE SCHOOL, CHARLE	EASTERN ILLINOIS UNIVERSITY ESTON, ILLINOIS	
M	ay 2012	
	YEAR	
I HEREBY RECOMMEND THAT TH THIS PART OF THE GRA	HIS THESIS BE ACCEPTED AS FULFILLING ADUATE DEGREE CITED ABOVE	
HHES H. MALLAGE 5/22/12 ESIS COMMITTEE CHAIR DATE	DEPARTMENT/SCHOOL CHAIR OR CHAIR'S DESIGNEE	5-Z
ESIS COMMITTEE MEMBER DATE	THESIS COMMITTEE MEMBER	DATE
ESIS COMMITTEE NEMBER DATE	THEOR COMMITTEE MEMBER	
117HA / OVEW # 5/22/12	2	

#### Abstract

This study was designed to examine the recruitment, matriculation, attrition, and retention behaviors of African American students attending the university. Data collected for this study were obtained from surveys completed by African American undergraduate students enrolled at the university either part-time or full-time at the time of the survey. Frequencies, percentages, and summaries were utilized in data analysis. There were also opportunities for participants to explain their responses.

The participants in this study completed a 25-item survey. This survey requested that participants rate (1) their level of involvement in co-curricular activities on campus; (2) their relationships with faculty and staff; (3) their overall satisfaction with the university; (4) their level of satisfaction with campus resources, (5) their overall academic ability and performance, (6) their emotional, physical, and mental health, and (7) their level of religiosity and spirituality. The results of this study suggest that African American students at the university were moderately involved in co-curricular activities on campus, had positive relationships with faculty and staff, demonstrated only minor unhealthy or risky behavior, were moderately involved in their faith, and were overall satisfied with the university and campus resources. The data also suggest that African American students at the university often experience moderate depression and feelings of being overwhelmed, yet choose to not utilize the counseling services provided on campus.

#### Acknowledgements

The completion of this project would not have been possible without the help of family, friends, faculty, and colleagues. I would like to take this opportunity to thank those who were involved in this process and showed me support throughout.

First, I would like to acknowledge my parents for all of the support they have given me throughout my educational endeavors. My mother, Sarah Rainer, and father, Willie Rainer, instilled in me the importance of education at a very young age, and they have always been there for me. For that, I am eternally grateful.

I am also profoundly grateful to have had the continuous moral, emotional, and spiritual support of my boyfriend, D'Wayne Jenkins. Throughout my entire time in the CSA program, D'Wayne has been there every step of the way helping me in any way he could. His countless words of encouragement when I felt discouraged will always be appreciated.

I offer my sincere appreciation to my thesis committee: Dr. James Wallace, Dr. Mona Davenport, and Dr. Angela Yoder. I acknowledge Dr. Wallace, my thesis chair, for all of the time he spent with me to ensure the success of this project. Without Dr. Wallace, this project would not have been possible. I am thankful for his commitment to this project.

Finally, I would also like to thank the members of my cohort who have helped during my time in the CSA program. Tynisa Collins, Tyson Holder, Levi Kosta-Mikel, and Andrea Bright have all contributed to my success in this program and to the success of my thesis. I acknowledge Levi for his assistance with the statistical aspects of this project. Without his help, I may not have finished my thesis before graduation.

# TABLE OF CONTENTS

ABSTRACT	i
ACKNOWLEDGEMENTS	ii
TABLE OF CONTENTS	iii-iv
LIST OF TABLES	v-vii
CHAPTER I	1
Introduction	1
Statement of the Problem	1
Statement of Purpose	3
Methodology	3
Research Questions	4
Limitations of the Study	4
Definitions of Terms	5
Research Sample	
Content Summary	6
CHAPTER 2	8
Review of Literature	8
Matriculation	8
Retention	8
Attrition	11
Graduation	14
CHAPTER 3	16
Methodology	16
Statement of the Purpose	16
Design of Study	16
Site Selection	16
Target Population	17
Data Collection	17
Instrumentation	18
Hypotheses	18
CHAPTER 4	19

Findings	19
Research Sample	
Demographic Data	
Research Question 1	
Research Question 2	
Research Question 3	
Research Question 4	
Research Question 5	
Research Question 6	
Research Question 7	
Summary of Data Analysis	
CHAPTER 5	
Discussion, Conclusions, and Recommendations	
Discussion	
Recommendations	
Conclusion	
REFERENCES	
APPENDICES:	
A. African American Student Survey  B. Instruments Modified for the African American Student Survey	76
C. Informed Consent.	108

# LIST OF TABLES

# **TABLE**

1	Frequencies and Percentages for Gender of African America Students20
2	Frequencies and Percentages for Age of African American Students
3	Frequencies and Percentages of Enrollment Status for African American Students20
4	Frequencies and Percentages for Class Standing of African American Students21
5	Frequencies and Percentages for Religious Preferences of African American Students21
6	Participants' Involvement in Co-Curricular Activities23
	Participated in Organized Presentations23
	Performed Volunteer Work23
	Voted in a Student Election24
	Participated in Student Government24
	Joined a Fraternity or Sorority25
	Played Varsity / Intercollegiate Athletics25
	Participated in Student Protests or Demonstrations26
	Participated in Student Clubs / Groups26
7	How Black Students Rated Their Relationships with Faculty and Staff28
	How Appealing are the Faculty and Staff?28
	Descriptions of Relationships with the Faculty (Professors and Class Instructors)28
	Descriptions of Relationships with Staff (Non-Faculty Employees)29
	Frequency that Students Sought Advice after Class29
	Extent that Students Communicated Regularly with Professors30
	Students' Perceptions of the Helpfulness of Faculty Members30
	Students' Perceptions of the Helpfulness of Staff Members31
8	Black Students' Perceptions of Their Overall Satisfaction with the University33
	How Students Rate their Appreciation for Difficult Subjects at the University33
	Students' Perceptions of Whether the University Prepares Students to Achieve Career  Goals33
	Students' Perceptions of Whether the University Prepares Students to Achieve Financial Security34
	How Students Rate their Overall Experiences at the University Thus Far34

	Students' Responses to Whether they would Recommend the University to Another Bi	<i>ack</i> 35
	Frequency that Students Felt Satisfied with the University	35
	Students' Responses to the Statement that they "Feel Like Dropping out of School"	
	Students' Responses to the Statement, "I Will Transfer to a Different Institution	
	because I Do Not Feel Welcome at the University"	36
9	Black Students' Satisfaction with the University Campus Resources	
	Frequency that Students Sought Personal Counseling	
	Students' Perceptions of the Help Received from the University Mentors	
	Students' Perceptions of the Help Received from University Tutors	
	Students' Perceptions of Help Received from Financial Aid	
	Students' Perceptions of Help Received from Academic Advisors	
10	How Black Students Rate their Overall Academic Ability and Performance	
	Students' Responses to the Statement, "I Should Be at a Higher Class Standing"	
	Students' Responses to Having Successfully earned 30 Credit Hours Each Year	
	Frequency that Students Arrived Late to Class	
	Frequency that Students Had Been on Academic Probation	
	How Students Rated Their Academic Ability	
	How Students Rated their Drive to Achieve	
	How Students Rated their Self-Confidence (Intellectual)	44
	How Students Responded to the Statement, "I Sometimes Feel that I Will Not	
	Graduate on Time"	45
	How Students Responded to the Statement, "I Feel Like Dropping Out of School"	45
11	How Black Students Rated their Own Emotional, Physical and Mental Health	
	Frequency that Students Felt Overwhelmed	
	Frequency that Students Felt Depressed	46
	Frequency that Students Sought Personal Counseling	47
	How Students Rated their Emotional Health	
	How Students Rated their Physical Health	47
12	How Black Students Rated their Level of Spirituality	
	Frequency that Students Attended a Religious Service Within the Past Year	

How Students Rated their Level of Religiousity	.49
How Students Rated their Level of Self-Understanding	.49
How Students Rated their Level of Spirituality	.50

#### **CHAPTER I**

#### Introduction

#### Statement of the Problem

There is a national crisis in the American university and college system regarding the status of African American students. Nationwide, the Black student college graduation rate is only 43 percent and Black women are far more likely than Black men to complete college. In 2006, the graduation rate of Black men was 36 percent, while the graduation rate for Black women was 47 percent. In contrast, the current graduation rate for white college students nationwide is 63 percent (Black, 2007). In 2009, only 19.4 percent of African Americans over the age of 25 held four-year college degrees (Vital, 2010). In 2008, only 32.6 percent of African Americans were enrolled in higher education (Vital, 2010). In 2011, the National Collegiate Athletic Association released graduation data for all students. The data presented showed that 43 percent of all Black students who matriculated at colleges and universities in 2004 earned their degrees within six years, while the six year rate for whites was 66 percent. By gender, the data showed that Black women graduated that same year at 46 percent, while the rate for White women was 68 percent. The graduation rate for Black men was 38 percent, while the rate for White men was 63 percent (JBHE, The Racial, 2011).

Not enough has been done to educate and graduate African American students as a whole. There is a demanding need for better, more pro-active and innovative academic programs for African American and other minority students on all college campuses to assist in their education, training, and degree attainment (Kemp, 1990).

Institutions of higher education are responsible for the healthy development and achievement of all students, regardless of race or ethnicity. When students are accepted into institutions of higher education, this responsibility lies with the university. Therefore, healthy

student development and successful matriculation must be evident in an institution's mission with evidence of multicultural initiatives (Lett, & Wright, 2003).

Williams and Leonard (1988) pointed out that many African American students display a pattern of taking courses outside their major requirements while dropping more difficult courses. They further stated that these students frequently have high grade point averages but do not graduate because of failure to fulfill course requirements in their selected majors. In addition, to gain entry into some majors, many universities require that a minimum grade point average be attained. As a result, too many African American students may be forced to select majors in which they lack true interest or may choose to leave the university out of frustration. This suggests that universities should develop and implement programs and services aimed at the education and progression of minority students toward graduation and social productivity rather than retention alone (Kemp, 1990).

There is no question that the retention of minority students is an important issue because without retention, graduation is impossible. However, since the mission of a higher education institution is to provide society with educated and productive citizens, to focus primarily on retention defeats this purpose. Theoretically, an undergraduate student could be retained in college indefinitely and never graduate (Kemp, 1990).

Solely emphasizing minority retention programs is insufficient for a number of reasons. First, retention programs, with regard to education, have historically been associated with underachieving, less prepared, slow, or learning disabled students. The use of this term, when preceded by the word 'minority' in place of a more suitable alternative (such as 'minority progression' or 'disposition') does a disservice to all students because of the associated negative connotations (Kemp, 1990).

A second reason that only focusing on retention programs is insufficient is the fact that African American and other minority college students do not want to simply be retained but educated, trained and graduated. African American students are just as motivated as are White students to be successful in college, but often feel, behave, and respond to situations in ways that frequently impede their academic progress and limit their academic success (Kemp, 1990; Nettles, 1988).

High attrition rates appear to be primarily caused by inferior K-12 preparation and an absence of a family college tradition, conditions that apply to a very large percentage of today's college-bound African Americans (Black, 2007). Nationally, universities have responded to the attrition dilemma with the creation of programs, committees, and offices aimed primarily at minority student retention. State and national conferences have been developed for college personnel to share ideas, research, and possible methods for dealing with the minority attrition problem (Kemp, 1990).

#### Statement of Purpose

The purpose of the present research was to examine the matriculation, attrition, and retention behaviors of African American students attending a midsize, Midwestern university. Specifically, the researcher was interested in developing a comprehensive understanding of the expectations, perceptions and actual experiences of African American undergraduates.

#### Methodology

The present study employed quantitative methodology for data collection through the use of an online questionnaire with response items based on Likert-type scales. The link to the questionnaire was sent to all African American students via their university email address. The

African American Student Survey features 30 questions: seven demographic inquiries, four queries regarding recruitment, two inquiries regarding matriculation, 10 queries regarding retention, three items regarding attrition, and four inquiries regarding graduation.

#### Research Questions

The seven research questions that guided the present study are as follows:

- I. To what extent were Black students involved in co-curricular activities?
- II. How did Black students rate their relationships with faculty and staff?
- III. How did Black students rate their overall satisfaction with the University?
- IV. How did Black students rate their level of satisfaction with campus resources?
- V. How did Black students rate their overall academic ability and performance thus far?
- VI. How did Black students rate their emotional, physical, and mental health?
- VII. How did Black students rate their spirituality?

#### Limitations of the Study

Response Rate. The response rate of the present study was only 0.06 percent (77 of 1142). Due to the low response rate, this study could be considered exploratory. Possible reasons for the low response rate include, but were not limited to, the following: potential participants were contacted via their university e-mail accounts and some students rarely check their accounts for new messages; some potential participants may no longer be students at the selected institution; and very few incentives were provided to potential participants. In addition, the timing of the survey could have affected the low response rate. The Principle Investigator (PI) administered the survey in the middle of the spring semester. By this time, many students

had likely already completed a number of other surveys for other graduate students and may have likely grown tired of participating in research studies.

Items. There seemed to be confusion surrounding certain items in the survey. This limitation was most noticeable in the participants' open-ended responses. For example, some participants may not have understood item 18, which asked students to describe their relationship with university staff (non-faculty paid employees). Some participants seemed to have not understood which individuals at the university were considered to be university staff. Based on some students' responses, some students thought that university staff was limited to custodians, maintenance workers, and food preparation workers. Some of the students' responses to specific items included: "BSWs are amazing"; "BSWs are very friendly and are concerned with how you are academically and personally"; "Don't really know too many, but the ones I have seen are pleasant"; "Don't understand the question"; "I don't interact with the staff, so I can't say anything good or bad"; "I don't know any"; "I'm not sure". Based on responses such as these, participants may not have known that university staff also includes employees in areas of Student Life, Greek Like, academic advising, TRiO supports services, Gateway, and Financial Aid.

Gender Ratios. Although more African American females were enrolled at the selected university in comparison to African American males at the time of the study, the ratio of males to females who participated in the study was far less than the ratio of males to females who were enrolled at the selected university. The PI was unable to determine why the male response rate was significantly less than the female response rate.

#### Definitions of Terms

Matriculation: The progression from one academic level to the next.

Retention: "the percent of entering students graduating or persisting in their studies at an institution" (Wyman, 1997, p. 29).

Attrition: "a longitudinal process of interactions between the individual and the academic and social systems of the college during which a person's experiences in those systems...continually modify his goals and institutional commitments in ways which lead to persistence and / or to varying forms of dropout" (Tinto, 1975, p. 94).

Graduation: The rate at which students graduate from the university within a four to six year period.

Black/African American: The "term used to designate people of African descent who are domiciled in the United States since 1865" (Asante, & Mazama, 2005, p.8).

Religiosity: "A belief in God accompanied by a commitment to follow principles believed to be set by God" (Vitell, Paolillo, & Singh, 2005, p. 175).

Spirituality: "Personal life principle [which] animates transcendent quality [of] relationship [with] God or god being" (Emblen, 1992, p. 45).

#### Research Sample

The research target population consisted of 1143 Black undergraduate students currently attending the selected university at least part-time. A total of 77 students completed the survey for a 0.06% response rate. Of the 77 survey participants, 60 (79%) were females and 17 (22%) were males (**Table 1**).

#### Content Summary

The present thesis is organized as follows. Chapter I contains the introduction, research hypotheses, definitions of terms, statement of purpose, limitations to the study, and a brief literature review of previously published work. Chapter II contains an extensive review of published literature on the experiences and perceptions of African American students regarding

their lives on PWI (predominantly White institution) campuses. Specific attention is focused on research studies that address issues of African American student recruitment, attrition, matriculation, graduation and retention behaviors. Chapter III includes the methodology used in data collection and analysis; an explanation of the instruments utilized in data collection (i.e., The CIRP Freshman Survey and the USG Telephone Graduates Survey). Chapter IV contains the findings from the survey instrument and is presented in narrative format along with attendant tables, graphs and charts reflecting participants' responses to the survey instruments. Chapter V is comprised of a discussion of the findings, conclusions drawn, and recommendations for both Student Affairs professionals and future research in this area.

#### **CHAPTER II**

#### **Review of Literature**

The focus of this chapter is to review previous research pertinent to the current state of the matriculation, retention, attrition, and graduation of African American students in the U.S.

#### Matriculation

African American students often feel alienated at PWIs and do not feel included in the college environment and community. As a result of these feelings, African American students may have problems matriculating successfully. African American students need support, inclusion, and acceptance in order to fit in the mainstream college environment. African American students, like any others, can excel in environments that promote inclusion and acceptance. When African American students are able to feel a part of an institution, they achieve at higher levels and successfully matriculate through their programs of study (Lett, & Wright, 2003; Swchwitzer, Griffin, Ancis, & Thomas, 1999).

Problems associated with African American matriculation revolve around "blatant and subtle barriers" related to discriminatory practices. According to Lett and Wright (2003) and Feagin (1992), barriers to obtaining a college education are inclusive but not limited to: a) alienation, b) isolation, c) racism, d) discrimination, e) intimidation, and f) problems associated with acquiring adequate financial aid.

#### Retention

Oliver, Rodriguez, and Mickelson (1985) reported that many programs and special services have been developed at PWIs to increase retention rates of African American students. However, these programs have often failed to prove effective in increasing retention and

graduation rates among African American students (Fortson, 1997). Cheatham, Tomlinson, and Ward (1990) reported that a major factor for many of these program failures is that they were developed without a clear understanding of the unique academic and psychosocial needs of African American students. Sedlacek (1987) identified eight variables that seem to be central to African American students' success, including positive self-concept, realistic self-appraisal, understanding of racism, ongoing community service, ability to indicate long-range goals, access to a strong support person, successful leadership experience, and knowledge acquired in an occupational field. If these variables are not present in programs and special services for African Americans at PWIs, then those programs and services may not be successful.

After surveying 206 African American students attending both PWIs and HBCUs (historically Black colleges and universities), Cokley (2000) found that African American students attending PWIs reported lower academic scores than did African American students attending HBCUs; even though those who attended PWIs entered college with higher high school grade point averages than did those who attended HBCUs. Cokley (2000) also found that African Americans who attended PWIs showed lower academic self-concept than did those who attended HBCUs. The results of these findings suggest a lack of support at PWIs for African American students which lead to low retention rates.

Researchers Fordham (1988), Hughes (1987), Lang (1986, 1992), Hudson, Henderson, and Henderson (2002) have found that undergraduate students' perceptions of campus climates significantly impact student retention on college campuses, and that the overall success of entering freshmen often depends on their first year of experience. They reported that poor advisement, having an undeclared major, lack of academic preparedness, and improper time management skills are several of the reasons that students do not succeed at the freshman level. The authors also reported that African American students' attrition and dropout rates increased

continuously, starting in the late 1980's. In a 2002 study, Hudson, Henderson, and Henderson investigated Florida A&M University's retention data submitted for the fall and spring semesters of 1997 through 2001. They found that although enrollment gradually increased through the years, the number of students who returned for the spring semesters declined, which resulted in lower retention rates. The authors concluded that financial difficulties, transitional issues, and family difficulties may have been factors contributing to the decline in retention.

Cabrera, Nora, Terenzini, Pascarella, and Hagedorn (1999) indicated that minority students tended to have more negative perceptions of campus racial and academic climates than did White students. Some researchers have reasoned that PWIs should do better in helping to meet African Americans' specific needs in areas such as financial aid, campus involvement, and academic performance to help improve African Americans' perceptions of campus climates (Furr, & Elling 2002; Gregory, 2000).

Research has shown that students' involvement in co-curricular activities and relationships with faculty and staff affect students' academic success on college campuses.

Lundberg and Schreiner (2004) found that students were more comfortable with faculty members of their own race / ethnicity. Astin (1984, 1993) found that students who were more involved in various aspects of college life, such as interaction with faculty and student organizations, had greater student learning and personal development than those who were not involved. Smedley et al. (1993) noted that certain factors (e.g., racism on campus and / or in the community, financial worries) threaten the effective adjustment for minority students, and they find it hard to concentrate on their studies and to trust faculty and administrators. Research has shown that some of the strongest predictors of college perseverance usually include; effective academic advising, interaction with faculty, living environment, classroom experience, and extracurricular activities (Levin, 1998). DeSousa and King (1992) reported that Black students were more

involved in organizations that consisted of predominantly Black students compared to White students in organizations like student government, resident hall associations, etc. The researcher also asserted that the students' involvement in largely Black organizations may provide a familiar cultural environment for Black students and help them establish social networks and support systems not found in the classroom atmosphere or residence halls (DeSousa, & King, 1992).

#### Attrition

Staying enrolled in school is a very difficult task for many African Americans. Fortson (1997) concluded that African American men endure one of the highest attrition rates of any other racial or gender group in higher education institutions. Fleming (1984) and Schwartz and Washington (2002) also explained that most African American men in higher education face significant challenges to attaining their degrees, resulting in higher attrition for Black males.

Negga, Applewhite, and Livingston (2007) found that the top five sources of stress at HBCUs and PWIs were (1) death of a family member (Intrapersonal Stress), (2) low grades (Academic Stress), (3) time management (Academic Stress), (4) boyfriend / girlfriend problems (Interpersonal Stress), and (5) missed classes (Academic Stress). These researchers also found that African American PWI students, in contrast with HBCU students, may need additional intervention or counseling services that are culturally sensitive to issues of racial discrimination, isolation and coping, as well as addressing the other factors common to all students at PWIs, such as control and self-esteem. According to Barksdale and Molock (2008), although rates of mental health problems among African Americans are similar to those of Caucasions, generally, African Americans are far less likely to seek mental health services. Their research suggests that African Americans prefer to seek mental help from non-mental health professionals, such as church members, family, or friends. The researchers reported that one prominent reason that

African Americans underutilize mental health care is the negative perceptions that family members and friends have toward professional health care.

Several studies have found that spirituality and religion can motivate students to succeed and remain in college (Donahoo, & Caffey, 2010). While Bryant and Astin (2008) suggested that students who struggle spiritually have poorer outcomes in college, Johnson et al (2003) and Walker and Dixon (2002) contended that African American students who demonstrate a high spiritual or religious orientation have higher grade point averages and earn more academic honors. These researchers also concluded that the support and encouragement students received from churches motivated them to stay and succeed in school. Donahoo and Caffey (2010) found that to varying degrees, students attributed their successful transitions into college, academic performance, career selection, ability to cope with stress, and the desire to accept and improve the lives of others to their church involvement, religious practice, and spirituality. As defined by Parks (2000), the church offers these students a form of community that promotes both their personal and religious development.

From their findings, Donahoo and Caffey (2010) concluded that although public institutions face constitutional limitations related to addressing the religious needs of their students, they should work with churches to help ensure that all interested students have access to the academic gains, coping skills development, and supports systems that religious and spiritual organizations have the ability to provide. They explained that this cooperation can include simple, inexpensive value neutral gestures such as allowing local churches to advertise their services in institutional publications, providing space during events devoted to recruiting students to participate in on and off-campus activities, and inviting churches to attend and participate in public events held on-campus. The authors found that this type of high-visibility involvement and access is especially crucial for African American students attending

predominantly white institutions as churches provide them with a place to escape much of the stress of campus life. From a retention perspective, student who attend church and experience the familial support provided motivates them to stay in school and pursue their degrees.

According to Donahoo and Caffey, the qualitative data obtained from several studies substantiates the fact that church involvement can help to promote student retention and college success by addressing a wide range of the spiritual, familial, social and personal needs of these students.

Researchers Fordham (1988), Hughes (1987), Lang (1986, 1992), Hudson, Henderson and Henderson (2002) have also noted that the overall success of entering freshmen oftentimes depends on their first year of experience. For example, they reported that poor advisement, having an undeclared major, lack of academic preparedness, and improper time management skills are several of the reasons that some students do not succeed at the freshman level. These researchers also reported that African American students' attrition and dropout rates increased continuously, starting in the late 1980's.

Researchers Perna (2006), St. John, Paulsen, and Carter (2005) and Titus (2006) have noted a relationship between inadequate financial support and student attrition. According to a study by Sallie Mae, the largest nonprofit provider of federal and private education loan funds in America, 69 percent of Blacks who enrolled in college but did not finish said that they left because of high student loan debt as opposed to 43 percent of White students who cited the same reason (Black, 2007). The availability of a high level of financial aid shields low-income Black students from financial pressures that may force minority students to leave college (Black, 2007).

In the African American community, lack of financial support may go unnoticed.

Research has shown that pride can impact African American males' ability to seek support

(Majors, & Billson, 1992). Majors and Billson characterized the pride that African American

males display as "cool pose" (p. 8); a façade used to display confidence and masculinity. They asserted that African American males adopt a cool pose to project control to others even when they are in need of assistance or guidance. High dropout rates appear to be primarily caused by inferior K-12 preparation and an absence of family college tradition, conditions that apply to a very large percentage of today's Black college-bound youth (Black, 2007).

#### Graduation

Provasnik and Shafer (2004) reported that, in 2001, 87.1% of all African American undergraduates attended PWIs while only 12.9% attended HBCUs. Their report indicated that PWIs accounted for 78.5% of all undergraduate degrees awarded to African American students nationally while HBCU's accounted for 21.5% all undergraduate degrees awarded to African American students. Currently, although a larger percentage of African Americans attend PWIs than attend HBCUs, yet PWIs account for a disproportionately lower percentage of degrees awarded to African American students.

Many PWI colleges and universities with high Black student graduation rates have established orientation and retention programs to help Black students adapt to the culture of predominantly White campuses. Mentoring programs for Black first-year students involving upperclassmen have been successful at many colleges and universities (Blount, 2011, pp. 22-29). Other institutions appear to improve graduation rates through strong Black student organizations that foster a sense of belonging among the Black student population. The presence or absence of these programs may have some impact on graduation rates (Black, 2007). Academic and social integration are said to directly influence success in college because, research shows that the more students are involved in the social and intellectual aspects of college life, and the more frequently

they interact with faculty and other students about learning issues, especially outside the classroom, the more students are likely to persist to degree completion (Astin, 1991).

The geographic location of colleges and universities plays a major role in Black student graduation rates. Colleges in rural areas with very small to negligible Black populations tend to have lower Black graduation rates than colleges and universities in urban areas (Black, 2007).

#### **CHAPTER III**

#### Methodology

Statement of Purpose

The purpose of the present research was to examine the matriculation, attrition, and retention behaviors of African American students attending a midsize, PWI located in the Midwestern U.S. Specifically, the researcher was interested in developing a comprehensive understanding of the expectations, perceptions and actual experiences of African American undergraduates attending the selected institution.

#### Design of Study

The purpose of the present research was to examine the recruitment, matriculation, attrition, and retention behaviors of African American students attending a midsize, public, PWI located in the Midwestern U.S. Specifically, the researcher was interested in developing a comprehensive understanding of the expectations, perceptions and actual experiences of African American undergraduates. In order to accomplish this task, quantitative inquiry in the form of a survey instrument was utilized for data collection and analysis. Specifically, the primary researcher modified components of the following published tools to fit the target population and specific institution: the USG Graduates Telephone Research Questionnaire from the University System of Georgia's African-American Male Initiative Research External Report, and The CIRP Freshman Survey from the Cooperative Institutional Research Program.

#### Site Selection

The present study was conducted at a four-year public institution located in the midwestern U.S. the selected university had a faculty-student ratio of 1 to 16. At the time of the current study, the university had a graduation rate of 59 percent.

#### Target Population

Enrollment figures for fall 2011 indicated that the university had a total enrollment of 11,630 with 1,382 (11.9%) self-identified African American undergraduate students. By class standing, 294 were seniors (120 men; 174 women); 295 juniors (106 men; 189 women); 316 sophomores (121 men; 195 women); and 490 freshmen (164 men; 326 women). Of the 1398 African American undergraduates, 947 were from the greater Chicago metropolitan area, 82 were from the East Saint Louis area, and 286 were from cities other than the Chicago / East St, Louis areas. The average student was between the ages of 18-24 years old.

#### Data Collection

The present study employed quantitative methodology for data collection through the use of an online questionnaire which predominantly contained questions answerable as Likert-type response scales. The link to the questionnaire was sent to all African American students via their institutional email address. In addition to demographic information, the survey addressed the following components of the undergraduate experience: initial contact with the university and recruitment personnel; classroom experiences; campus climate; social norms; religious participation; perceptions of themselves (i.e., degrees of satisfaction with their emotional and mental health); involvement with faculty and staff (e.g., mentoring); co-curricular activities (e.g., athletics, interactions with community businesses, cross-cultural interactions, RSOs); access to and utilization of campus resources (e.g., tutoring, financial aid); and, awareness of programs and offices specifically established to provide services and fulfill the needs of the minority or African American populations. In addition, consideration was given to degree specific programs

(e.g., number of Black students enrolled in pre-Nursing and other programs that required a brief campus stay and a similar time spent on a satellite campus or office).

#### Instrumentation

The institutional African American Student Survey was a mixture of items developed inhouse by the PI, in addition to modified items from the USG Graduates Telephone Research Questionnaire and the The CIRP Freshman Survey. The USG Graduates Telephone Research Questionnaire was obtained from *The University System of Georgia's African-American Male Initiative Research External Report On Attitudes and Barriers Impacting the Participation of African-American Males in the University System of Georgia*, Matlock Advertising and Public Relations and Paul A. Warner Associates. The CIRP Freshman Survey was obtained from the Higher Education Research Institute, home of the Cooperative Institutional Research Program, and this version was administered in 2005. The institutional African American Student Survey used in the present study features 30 questions: seven demographic inquiries, four queries regarding recruitment, two inquiries regarding matriculation, 10 queries regarding retention, three items regarding attrition, and four inquiries regarding graduation.

#### Hypotheses

- 1. Black students will report being dissatisfied with access to campus resources.
- 2. Black students will report having absent or negative relationships with their instructors.
- 3. Black students will report having strong feelings of isolation and alienation.
- 4. Black students will report being dissatisfied overall with the university as an institution.

#### **CHAPTER IV**

#### **Findings**

The purpose of the present study was to examine the perceptions and experiences of Black students attending a midsize, public university located in the midwestern U.S. regarding their matriculation, attrition, and retention behaviors. The research data in this chapter are responses to a survey completed by Black students enrolled at the selected institution.

#### Research Sample

The research sample consisted of 1143 Black students currently attending at least part-time at the selected university. A total of 77 students completed the survey for a 0.06% response rate. Of the 77 survey participants, 60 (79%) were women and 17 (22%) were men (**Table 1**).

# Demographic Data

Tables 1 thru 5 below are displays of the demographic profiles of study participants.

Specifically, Table 1 shows the gender of the participants, the majority of whom were females.

On the basis of gender, demographics are statistically significant yet cannot be generalized to the entire Black student population due to the low number of male participants. Table 2 shows the ages of the participants which ranged from 18-24 years; the majority of the participants were 21 years old. Table 3 shows the enrollment status of the participants. With the exception of two, all participants in this survey were fulltime students. Table 4 shows the class standing of the participants; the majority of whom were seniors. Table 5 shows the religious preferences of the participants; the majority of the participants reported practicing the Baptist faith.

Table 1

Frequencies and Percentages for Gender of African American Students

Gender	N	%	
Female	60	77.9	
Male	17	22.1	
Total	77	100.0	

Table 2

Frequencies and Percentages for Age of African American Students

Age	Male		Female	
	N	%	n	%
18	1	6.3	9	15.5
19	1	6.3	10	17.3
20	3	18.7	8	13.8
21	4	25.0	24	41.4
22	3	18.7	5	8.6
23	1	6.3	1	1.7
24	3	18.7	1	1.7
Total	16	100.0	58	100.0

Table 3

Frequencies and Percentages of Enrollment Status for African American Students

Enrollment Male				Female	
Status	N	%	n	%	
Full-time	17	100.0	58	96.7	
Part-time	0	0.0	2	3.3	
Total	17	100.0	60	100.0	

Table 4

Frequencies and Percentages for Class Standing of African American Students

Class Standing	Male		Female	
	n	%	n	%
Freshman	3	17.6	11	18.3
Sophomore	1	5.9	13	21.7
Junior	5	29.4	17	28.3
Senior	8	47.1	19	31.7
Total	17	100.0	60	100.0

Table 5

Frequencies and Percentages of Religious Preferences for African American Students

Religious	Male		Female		
Preference	N	%	n	%	
Baptist	3	17.7	15	25.0	
Buddhist	1	5.9	0	0.0	
Church of Christ	2	11.7	11	18.3	
Islamic	0	0.0	1	1.7	
Lutheran	0	0.0	2	3.3	
Methodist	1	5.9	1	1.7	
Presbyterian	1	5.9	1	1.7	
Roman Catholic	1	5.9	5	8.3	
United Church of			3	5.0	
Christ/Congregational	1	5.9			
Other Christian	5	29.4	12	20.0	
Other Religion	0	0.0	3	5.0	
No Religious Preference					
or Affiliation	2	11.7	6	10.0	
Total	17	100.0	60	100.0	

#### Research Questions

#### Research Question 1

Research question one was designed to ascertain the extent to which students were involved in co-curricular activities on campus at the university. Participants were instructed to indicate their responses on Likert-type scales. Responses to survey items 22.3, 22.10, 22.12, 22.14, 22.15, 22.16, 22.17, and 22.23 were assessed to determine how involved students were in co-curricular activities on campus. Survey responses showed that the majority of students (n=35or 45.5%) sometimes participated in organized presentations on campus. The majority of students (n = 37 or 48.1%) sometimes performed volunteer work on campus. The same number of participants (n = 22 or 28.6%) indicated that they never or sometimes voted in a student election. The majority of students (n = 54 or 70.1%) never participated in student government activities. The majority of students (n = 64 or 83.1%) additionally never joined a fraternity or sorority. The majority of students (n = 60 or 77.9%) never played on a varsity / intercollegiate athletic team. The majority of students (n = 56 or 73.7%) never participated in student protests or demonstrations. A large minority of students (n = 27 or 35.5%) always participated in Registered Student Organizations. The frequencies and percentages for individual participant responses can be found in Table 6.

Table 6

Participants' Involvement in Co-Curricular Activities

Participated in Organized Presentations

Item 22.3	n	%	
Never	3	3.9	
Almost Never	5	10.4	

# Never 3 3.9 Almost Never 5 10.4 Sometimes 35 45.5 Almost Always 27 35.1 Always 7 9.1

# Performed Volunteer Work

Item 22.10	n	%	
Never	4	5.3	
Almost Never	12	15.8	
Sometimes	37	48.7	
Almost Always	14	18.4	
Always	9	11.8	
Missing	1		

# Voted in a Student Election

Item 22.12	n	%	1.010
Never	22	28.6	
Almost Never	13	16.9	
Sometimes	22	28.6	
Almost Always	14	18.2	
Always	6	7.8	

# Participated in Student Government

Item 22.3	n	%	
Never	54	70.1	
Almost Never	15	19.5	
Sometimes	3	3.9	
Almost Always	3	3.9	
Always	2	2.6	

# Joined a Fraternity or Sorority

Item 22.15	n	%	
Never	64	83.1	
Almost Never	2	2.6	
Sometimes	2	2.6	
Almost Always	1	1.3	
Always	8	10.4	

# Played Varsity / Intercollegiate Athletics

Item 22.3	n	%	
Never	60	78.9	
Almost Never	5	6.6	
Sometimes	2	2.6	
Almost Always	1	1.3	
Always	8	10.5	
Missing	1		

# Participated in Student Protests or Demonstrations

Item 22.17	n	%	
Never	56	73.7	
Almost Never	11	14.5	
Sometimes	7	9.2	
Almost Always	1	1.3	
Always	1	1.3	
Missing	1		

# Participated in Student Clubs/Groups

Item 22.23	n	%	
Never	11	14.5	
Almost Never	4	5.3	
Sometimes	17	22.4	
Almost Always	17	22.4	
Always	27	35.5	
Missing	1		

## Research Question 2

Research question two was designed to determine how Black students rated their relationships with faculty and staff. Participants were instructed to indicate their responses on a Likert-type scale. Responses to survey items 13.11, 16, 17, 18, 19, 22.11, 22.21, 22.5, and 22.6 were analyzed to determine how Black students rated their relationships with faculty and staff. For items 16 and 18, participants were instructed to explain their responses. The majority of students (n = 27 or 35.1%) responded that they agree that the faculty and staff were appealing. The majority of students (n = 42 or 54.5%) indicated that they have a positive relationship with the university faculty (professors and instructors). The majority of students (n = 32 or 41.6%) also indicated that they had a positive relationship with the university staff (non-faculty employees). The majority of students (n = 27 or 35.5%) indicated that they sometimes asked professors for advice after class and (n = 28 or 36.4%) indicated that they sometimes communicate regularly with their professors. The majority of students (n = 34 or 44.2%) found the university faculty to be helpful and at least (n = 61 or 74.6%) of participants found the university staff to be helpful or very helpful. The frequency and percentages for individual participant responses can be found in **Table 7**. For items 17 and 19, students were also asked to explain their responses to items 16 and 18. A summary of their responses are included below.

Table 7

How Black Students Rate Their Relationships with Faculty and Staff

How Appealing are the Faculty and Staff?

Item 13.11	n	%	
Not Appealing	6	7.8	
Somewhat Appealing	10	13.0	
Neutral	20	26.0	
Appealing	27	35.1	
Very Appealing	14	18.2	

## Descriptions of Relationships with Faculty (Professors and Class Instructors)

Item 16	n	%	
Extremely Negative	0	0	
Negative	1	1.3	
Neutral	23	29.9	
Positive	42	54.5	
Extremely Positive	11	14.3	

# Descriptions of Relationships with Staff (Non-Faculty Employees)

Item 18	n	%	
Extremely Negative	1	1.3	
Negative	1	1.3	
Neutral	32	41.6	
Positive	28	36.4	
Extremely Positive	15	19.5	

# Frequency that Students Sought Advice after Class

Item 22.11	n	%	
Never	7	9.2	
Almost Never	7	9.2	
Sometimes	27	35.5	
Almost Always	18	23.7	
Always	17	22.4	
Missing	1		

# Extent that Students Communicated Regularly with Professors

Item 22.21	n	%	18.00118.00.1.1.1.1.1.1.1.1.1.1.1.1.1.1.
Never	5	6.5	
Almost Never	3	3.9	
Sometimes	28	36.4	
Always	27	35.1	
Almost Always	14	18.2	

# Students' Perceptions of the Helpfulness of Faculty Members

Item 23.5	n	%	
Not Helpful	0	0	
Slightly Helpful	6	7.8	
Neutral	10	13.0	
Helpful	34	44.2	
Very Helpful	27	35.1	

Students' Perceptions of the Helpfulness of Staff Members

Item 23.6	n	%	
Not Helpful	0	0	
Slightly Helpful	8	10.7	
Neutral	11	14.7	
Helpful	28	37.3	
Very Helpful	28	37.3	

Students' Explanations of Their Responses to Item 16

For item 17, students were asked to explain their responses to item 16, which asked students to describe their relationships with faculty members. Mostly, student responses indicated that they had positive and engaging relationships with their faculty members. One student wrote, "All of my teachers are very understanding, the [university] faculty really care about us students and take a divine interest in us somewhat like we are all family." Another student responded, "I am very close with many of my professors." A third participant responded, "I love all my teachers they're awesome." A large number of responses showed neutral relationships with professors. Although there were more positive responses, there were some negative responses. One negative response was, "I am not close to any professors." Another participant wrote, "I don't feel like they try to interact with students. It's very dry and bland. I always feel as though they're just doing their job and nothing more. One thing I appreciate is when you can tell when your professors truly want their best interest for your success as a student and person. Have not seen that on campus..."

Students' Explanations of Their Responses to Item 18

For item 19, students were asked to explain their responses to item 18, which asked students to describe their relationships with university staff members. Largely, students

responded that they did not know any staff members or had never interacted with staff members. Some students replied that they were confused by the question. Other students answered the question generally, such as, "Everyone I have come in contact with at [the university] has been pleasant and polite." Additional responses showed that some students believed that "staff" meant service workers such as custodians and maintenance workers. This is shown in the response, "The BSWs and maintenance people always speak and are very kind."

## Research Question 3

Research question three was designed determine how Black students rated their overall satisfaction with the university. Participants were instructed to indicate their responses on a Likert-type scale. Survey items 13.1, 13.8, 13.9, 14, 15, 20, 21 22.19, 25.2, 25.4 were combined to determine how Black students rate their overall satisfaction with various components of their experience at the university. The majority of students (n = 28 or 36.4%) strongly disagreed that they enjoy the difficult subjects that they have taken at the university. Most students strongly agreed that the university prepares students to achieve financial security (n = 40 or 51.9%) and career goals (n = 46 or 59.7%). The majority of students described their overall experience as positive (n = 42 or 54.5%) while only two students (2.6%) described their overall experiences as negative or extremely negative. A large majority of students (n = 68 or 88.3%) indicated that they would recommend the university to another African American student, and most students (n = 32 or 41.6%) indicated that they were almost always satisfied with the university; only two students (2.6%) admitted that they never were satisfied with the university. An overwhelming 57.9% (n = 44) of students strongly disagreed that they felt like dropping out of school, while the least amount of students (n = 3 or 3.9%) indicated that they strongly agree that they felt like dropping out of school. The majority of students (n = 47 or 63.5%) responded that they strongly

disagree that they will transfer to a different institution due to feeling unwelcome at the university. The frequency and percentages for individual participant responses can be found in **Table 8**. For items 15 and 21, students were also asked to explain their responses to items 14 and 20. A summary of their responses are included below.

Table 8

Black Students' Perceptions of their Overall Satisfactions with the University

How Students Rated their Appreciation for Difficult Subjects at the University

Item 13.1	n	%	
Strongly Disagree	28	36.4	
Disagree	16	20.8	
Neutral	13	16.9	
Agree	17	22.1	
Strongly Agree	3	3.9	

Students' Perceptions of Whether the University Prepares Students to Achieve Career Goals

n	%	
0	0	
0	0	
8	10.4	
23	29.9	
46	59.7	
	0 0 8 23	0 0 0 0 8 10.4 23 29.9

Students' Perceptions of Whether the University Prepares Students to Achieve Financial Security

n	%	
1	1.3	
4	5.2	
9	11.7	
23	29.9	
40	51.9	
	1 4 9 23	1 1.3 4 5.2 9 11.7 23 29.9

How Students Rated their Overall Experiences at the University Thus Far

n	%	
1	1.3	
1	1.3	
17	22.1	
42	54.5	
16	20.8	
	1 1 17 42	1 1.3 1 1.3 17 22.1 42 54.5

Students' Responses on Whether They Would Recommend the University to Another Black Student

Item 20	n	%	
Yes	68	88.3	
No	9	11.7	

## Frequency that Students Felt Satisfied with the University

Item 22.19	n	%	
Never	2	2.6	
Almost Never	4	5.2	
Sometimes	22	28.6	
Almost Always	32	41.6	
Always	17	22.1	

## Students' Responses to the Statement that They "Feel Like Dropping Out of School"

Item 25.2	n	%	
Strongly Disagree	44	57.9	
Disagree	15	19.7	
Neutral	8	10.5	
Agree	6	7.9	
Strongly Agree	3	3.9	
Missing	1		

Students' Responses to the Statement, "I will transfer to a different institution because I do not feel welcome at the University."

Item 25.4	n	%	
Strongly Disagree	47	63.5	
Disagree	15	20.3	
Neutral	7	9.5	
Agree	3	4.1	
Strongly Agree	2	2.7	

## Students' Explanations of Their Responses to Item 14

For item 15, students were asked to explain their responses to item 14, which asked students to describe their overall experiences at the university. Generally, students were pleased with their experiences. Many students indicated that the students and faculty at the university are friendly. For example, one student wrote, "Almost everyone is friendly and [the university] has a laid-back vibe to it, it's great..." Another student wrote, "Every day I am meeting someone new, the atmosphere is great, and I love being here." Students also noted they felt that they had grown as a person since their start at attending [the university]. For instance, one participant wrote, "I have grown into a better person and student since I have been here." Another student wrote, "I have had some wonderful and life changing experiences at [the university], I love the school, and have grown to adapt to the environment..." A third student wrote, "I have learned a lot and grown even more as a person." There were numerous responses about being pleased with the diverse activities and organizations offered to students on campus. One student stated, "I have been involved with over 10 organizations on campus. Each are very diverse and have helped me throughout the school year socially and academically." A second participant wrote, "I have had several leadership positions on campus and have maintained a steady high gpa while

meeting various types of people." Another student stated, "There are a variety of diverse programs, events, and workshops that make a difference on the individuals on a college campus." In contrast, some students displayed a negative overall view of [the university]. For example, one student noted, "I am not comfortable with certain individuals on campus. I feel as though I am unwanted because of my racial background." Another student responded, "If given the chance, I would choose another school." A third student stated, "The school has not been enjoyable the police harrass black students."

Students' Explanations of Their Responses to Item 20

For item 21, students were asked to explain their responses to item 20, which asked students if they would recommend the university to another African American student? Generally, students' responses were positive. While there is no way to determine exactly how each student responded to item 20, positive responses for item 21 indicates that those students would recommend the university to another African American student. In responding to item 21, most students regarded the university as a diverse, welcoming institution. For example, one student wrote, "The university is a diverse environment for you to grow as a person and enhance as a race." Another participant responded, "The university is a school of diversity that focuses on your success." A third student responded, "It's a diverse campus, so you'll get to meet lots of new people and take your first step into a larger world." Some students responded that they enjoyed the close-knit, family-like feel of the university campus. One participant wrote, "African American students are friendly and because there aren't many of us here, there is a close connection..." Another student wrote, "I would tell them that there is a close knit connection between the African Americans and the university is a great school." Students also noted that they were pleased with the number and quality of organizations on campus designed for African Americans. One participant stated, "... There are plenty of organizations to join in

order to feel welcomed, accepted, and at "home". Another student responded, "There are several organizations here at [the university] that strive to see the success and growth of its African Americans." Some students, however, did respond negatively. One student replied, "I would not recommend another African American because this city is too rural and some of the people living in this community aren't as accepting towards different cultures and ethnicities."

"Another student wrote, "This university does not meet the needs of African American students."

## Research Question 4

Research question four was designed determine how Black students rated their level of satisfaction with the university's campus resources. Participants were instructed to indicate their responses on a Likert-type scale. Survey items 22.20, 23.1, 23.2, 23.3, and 23.4 were joined to determine how Black students rate their satisfaction with university campus resources. A vast majority of students (n = 47 or 61.0%) responded that they had never sought personal or mental health counseling. Whereas some students indicated that university mentors (n = 24 or 31.2%) and tutors (n = 23 or 29.9%) were helpful, respectively, the majority responded that university academic advisors (n = 30 or 39.0%) and Financial Aid (n = 38 or 49.4%) were very helpful. The frequency and percentages for individual participant responses can be found immediately following in **Table 9**.

Table 9

Black Students' Satisfaction with University Campus Resources

Frequency that Students Sought Personal Counseling

Item 22.20	n	%	
Never	47	61.0	
Almost Never	8	10.4	
Sometimes	15	19.5	
Always	3	3.9	
Almost Always	4	5.2	

Students' Perceptions of the Help Received from University Mentors

Item 23.1	n	%	
Not Helpful	13	16.9	
Slightly Helpful	8	10.4	
Neutral	19	24.7	
Helpful	24	31.2	
Very Helpful	13	16.9	

Students' Perceptions of the Help Received from University Tutors

Item 23.2	n	%	
Not Helpful	8	10.4	
Slightly Helpful	7	9.1	
Neutral	21	27.3	
Helpful	23	29.9	
Very Helpful	18	23.4	

Students' Perceptions of Help Received from Financial Aid

Item 23.3	n	%	
Not Helpful	4	5.2	
Slightly Helpful	5	6.5	
Neutral	9	11.7	
Helpful	21	27.3	
Very Helpful	38	49.4	

Students' Perceptions of the Help Received from University Academic Advisors

Item 23.4	n	%	
Not Helpful	6	7.8	
Slightly Helpful	6	7.8	
Neutral	7	9.1	
Helpful	28	36.4	
Very Helpful	30	39.0	

## Research Question 5

Research question five was designed determine how Black student rated their overall academic ability and performance thus far. Participants were instructed to indicate their responses on a Likert-type scale. Survey items 21.1, 12.2, 22.13, 22.25, 24.1, 24.6, 24.13, 25.1, and 25.2 were combined to determine how Black students view their own overall academic ability and performance thus far. Although participants' responses varied a great deal, most students (n = 21 or 27.3%) indicated that they agree that they should have been at a higher class standing than they were at the time of the survey. Students' responses show that the majority of students strongly agree (n = 28 or 36.8%) or agree (n = 21 or 27.6%) that they have had no problems earning thirty credit hours per academic year. Most students responded that they are almost never late to class (n = 31 or 40.3%), and have never been on academic probation (n = 61or 80.3%). The majority of students (n = 44 or 57.1%) rated their own academic ability as above average. Most students indicated that their drive to achieve was either in the highest 10% (n =34 or 44.2%) or above average (n = 26 or 33.8%). Most participants rated their own intellectual self-confidence as in the highest 10 percent (n = 29 or 37.7%) or above average (n = 27 or 35.1%). Participants' responses to item 25.1 were somewhat split, but the largest minority of students (n = 21 or 27.6%) indicated that they strongly disagree that they sometimes felt as though they would not graduate on time. An overwhelming majority of participants (n = 44 or 57.9%) responded that they strongly disagree that they felt like dropping out of school. frequency and percentages for individual participant responses can be found in Table 10.

Table 10

How Black Students Rate their Overall Academic Ability and Performance

Students' Responses to the Statement, "... I Should Be at a Higher Class Standing"

n	%	
15	19.5	
18	23.4	
15	19.5	
21	27.3	
8	10.4	
	15 18 15 21	15 19.5 18 23.4 15 19.5 21 27.3

Students' Responses to Having Successfully earned 30 Credit Hours Each Year

Item 12.2	n	%	
Strongly Disagree	3	3.9	
Disagree	9	11.8	
Neutral	15	19.7	
Agree	21	27.6	
Strongly Agree	28	36.8	
Missing	1		

Frequency that Students Arrived Late to Class

Item 22.13	n	%	
Never	9	11.7	
Almost Never	31	40.3	
Sometimes	28	36.4	
Almost Always	8	10.4	
Always	1	1.3	
Aiways	I	1.5	

# Frequency that Students Had Been on Academic Probation

Item 22.25	n	%	
Never	61	80.3	
Almost Never	10	13.2	
Sometimes	4	5.3	
Almost Always	0	0	
Always	1	1.3	
Missing	1		

## How Students Rated their Academic Ability

n	%	
0	0	
0	0	
23	29.9	
44	57.1	
10	13.0	
	0 0 23 44	0 0 0 0 23 29.9 44 57.1

How Students Rated their Drive to Achieve

Item 24.6	n	%	
Lowest 10%	0	0	
Below Average	1	1.3	
Average	16	20.8	
Above Average	26	33.8	
Highest 10%	34	44.2	

# How Students Rated their Self-Confidence (Intellectual)

Item 24.13	n	%	
Lowest 10%	1	1.3	
Below Average	1	1.3	
Average	19	24.7	
Above Average	27	35.1	
Highest 10%	29	37.7	

How Students Responded to the Statement, "I Sometimes Feel that I Will Not Graduate On Time"

n	%	
21	27.6	
16	21.1	
15	19.7	
17	22.4	
7	9.2	
1		
	21 16 15 17	21 27.6 16 21.1 15 19.7 17 22.4 7 9.2

How Students Responded to the Statement, "I Feel Like Dropping Out of School"

Item 25.2	n	%	
Strongly Disagree	44	57.9	
Disagree	15	19.7	
Neutral	8	10.5	
Agree	6	7.9	
Strongly Agree	3	3.9	
Missing	1		

## Research Question 6

Research question six was designed to determine how Black students rate their own emotional, physical, and mental health. Participants were instructed to indicate their responses on a Likert-type scale. Survey items 22.8, 22.9, 22.20, 24.7, and 24.10 were linked to determine how Black students assessed their own emotional, physical, and mental health. Most students

either sometimes (n = 38 or 51.4%) or almost always (n = 26 or 35.1%) felt overwhelmed with all they had to do. The majority of students (n = 33 or 42.9%) sometimes felt depressed. An overwhelming majority of students (n = 47 or 61.0%) never sought personal counseling. The largest minority of students regarded their emotional health (n = 28 or 36.8%) and their physical health (n = 35 or 45.5%) as average. The frequency and percentages for individual participant responses can be found in **Table 11**.

Table 11

How Black Students Rated their Own Emotional, Physical, and Mental Health

Frequency that Students Felt Overwhelmed

Item 22.8	n	%	
Never	2	2.7	
Almost Never	4	5.4	
Sometimes	38	51.4	
Almost Always	26	35.1	
Always	4	5.4	

## Frequency that Students Felt Depressed

Item 22.9	n	%	
Never	16	20.8	
Almost Never	18	23.4	
Sometimes	33	42.9	
Almost Always	6	7.8	
Always	4	5.2	

# Frequency that Students Sought Personal Counseling

Item 22.20	n	%	
Never	47	61.0	
Almost Never	8	10.4	
Sometimes	15	19.5	
Almost Always	3	3.9	
Always	4	5.2	

## How Students Rated their Emotional Health

Item 24.7	n	%	
Never	3	3.9	
Almost Never	3	3.9	
Sometimes	28	36.8	
Almost Always	19	25.0	
Always	23	30.3	
Missing	1		

# How Students Rated their Physical Health

Item 24.10	n	%	
Lowest 10%	2	2.6	
Below Average	8	10.4	
Average	35	45.5	
Above Average	20	26.0	
Highest 10%	12	15.6	

## Research Question 7

Research question seven was designed to determine how Black students rated their level of spirituality. Participants were instructed to indicate their responses on a Likert-type scale. Survey items 22.1, 24.12, 24.14, and 24.15 were combined to determine how students assessed their spirituality. Most students (n = 34 or 44.2%) responded that they sometimes attended a religious service within the past year. The majority of students (n = 33 or 42.9%) assessed their level of religiousness as average, their level of self-understanding as above average (n = 31 or 40.8%), and their level of spirituality as either average (n = 30 or 39.0%) or above average (n = 28 or 36.4%). The frequency and percentages for individual participant responses can be found in **Table 12.** 

Table 12

How Black Students Rated their Level of Spirituality

Frequency that Students Attended a Religious Service Within the Past Year

Item 22.1	n	%	
Never	16	20.8	
Almost Never	10	13.0	
Sometimes	34	44.2	
Almost Always	9	11.7	
Always	8	10.4	

How Students Rated their Level of Religiosity

Item 24.12	n	%	
Lowest 10%	5	6.5	
Below Average	6	7.8	
Average	33	42.9	
Above Average	20	26.0	
Highest 10%	13	16.9	

# How Students Rated their Level of Self-Understanding

Item 24.14	n	%	
Lowest 10%	1	1.3	
Below Average	2	2.6	
Average	16	21.1	
Above Average	31	40.8	
Highest 10%	26	34.2	
Missing	1		

How Students Rated their Level of Spirituality

Item 24.15	n	%	
Lowest 10%	1	1.3	
Below Average	1	1.3	
Average	30	39.0	
Above Average	28	36.4	
Highest 10%	17	22.1	

## Summary of Data Analysis

Included above are students' responses to survey inquiries regarding matriculation, retention, attrition, and graduation. This study was designed to gain an understanding of the experiences and perceptions of Black students at the university, paying close attention to matriculation, retention, attrition, and graduation rates. Frequencies and percentages, along with some open-ended responses, were used to note the importance of the categories of involvement in co-curricular activities, relationships with faculty and staff, overall satisfaction with the university, satisfaction with campus resources, the extent of unhealthy or risky behaviors, overall academic ability and performance, emotional, physical, and mental health, and level of spirituality.

The result s of this study showed that the majority of students had positive perceptions of their experiences at the university. Overall, the current research sample was somewhat involved in co-curricular activities on campus at the university. Most students showed to have positive relationships with faculty and staff members. The majority of the participants indicated that they were, overall, satisfied with the university and campus resources, including mentors, tutors, and Financial Aid, although most had never utilized the personal counseling resources. Generally, students exhibited little to no unhealthy or risky behaviors during the past year. Largely, the

participants had a positive view of their overall academic ability and performance thus far. Most students reported moderate to poor emotional, physical, and mental health. The participants' responses in research question eight showed moderate levels of religiosity and spirituality.

As shown in the tables above, some participant(s)' responses are missing from this study. It is not possible to know who or why one individual or multiple individuals did not respond to certain questions, as I do not have individual participants' contact information.

#### **CHAPTER V**

## Discussion, Conclusions, and Recommendations

The purpose of the present study was to examine the perceptions and experiences of Black students at the university with a focus on their matriculation, attrition, and retention behaviors. This chapter includes a discussion, recommendations for future research, and conclusions.

#### Discussion

Research question one asked participants to determine how involved they were in cocurricular activities on campus. According to Astin (1984, 1993), students who are more
involved in various aspects of college life, like interaction with faculty and student organizations,
had greater student learning and personal development than those who were not involved. In
addition, DeSousa and King (1992) reported that students' involvement in largely Black
organizations may provide a familiar cultural environment for Black students and help them
establish social networks and support systems not found in the classroom atmosphere or
residence halls.

The present study showed that a low number of participants voted in student elections, and most students indicated that they have never participated in student government activities or student protests or demonstrations. In contrast, many students reported that they had always or almost always participated in organized presentations and registered student organizations. Based on students' open-ended responses to item 21, many students are involved in organizations that are predominantly Black. Thus, the university likely retained most of the participants in this study since they have established social networks and support systems, and they likely matriculated and graduated successfully.

Research question two asked students to rate their relationships with faculty and staff. Levin (1998) reported that some of the strongest predictors of college perseverance usually include interaction with faculty. Smedley et al. (1993) noted that certain factors (racism on campus and/or in the community, financial worries) threaten the effective adjustment for minority students, and they find it hard to concentrate on their studies and to trust faculty and administration. Astin (1984, 1993) found that students who have more interaction with faculty had greater student learning and personal development than those who did build relationships with faculty members. In the present study, the majority of participants indicated that they had a positive relationship with university faculty (professors and class instructors), sometimes asked them for advice after class, and frequently communicated with them. Overall, students also found staff members to be either helpful or very helpful. Therefore, although the vast majority of faculty members at the university are White, the Black students who comprised the sample in this study indicated that they have very positive relationships and frequent interaction with faculty members. Thus, according to previous research, the students in the present study likely produced a high retention rate and likely matriculated and graduated successfully.

Research questions three and four asked participants to rate their overall satisfaction with the university and campus resources. Generally, students indicated that the university had been successful at preparing them to achieve financial security and their career goals, described their overall experience at the university as positive, and would recommend the university to another prospective Black student. Researchers Fordham (1988), Hughes (1987), Lang (1986, 1992), Hudson, Henderson, and Henderson (2002) found that undergraduate students' perceptions of campus climate significantly impact student retention on college campuses. When African American students are able to feel a part of an institution, they achieve at higher levels and successfully matriculate through their programs of study (Lett, & Wright, 2003; Swchwitzer,

Griffin, Ancis, & Thomas, 1999). The results of the present study show that the participants generally had positive perceptions of the campus climate and were likely to successfully matriculate and graduate.

Research question five asked students to rate their overall academic ability and performance. According to Cokley (2000), found that African American students attending PWIs reported lower academic scores than did African American students attending HBCUs; even though those who attended PWIs entered college with higher high school grade point averages than did those who attended HBCUs. Cokley (2000) also found that African Americans who attended PWIs showed lower academic self-concept than did those who attended HBCUs. The results of these findings suggest a lack of support at PWIs for African American students which lead to low retention rates. In the present study, although most students agreed that they should have been at a higher class standing then they were at the time of the survey, the participants usually indicated a high self-concept of academic ability and performance, rating their own academic ability as above average and their own intellectual self-confidence as in the highest 10 percent. They also responded that they had never been on academic probation. These results suggest that Black students at the selected university had a strong academic support system on campus.

Research question six asked student to rate their own emotional, physical, and mental health. Negga, Applewhite, and Livingston (2007) found that African Americans cope with a variety of stressors at both HBCUs and PWI's, while the most frequent source of stress found was academic stress. In addition, the researchers reported that African American PWI students are often subject to issues of racial discrimination, isolation and coping, as well as addressing the other factors common to all students at PWIs, such as control and self-esteem. However, Barksdale and Molock (2008) reported that, generally, African Americans are far less likely than

Caucasians to seek mental health services due to negative perceptions of counseling from others, and that African Americans prefer to seek mental help from non-mental health professionals, such as church members, family, or friends. In the present study, participants indicated that they had been overwhelmed often, had sometimes felt depressed, and rated their emotional and physical health as average. However, the majority of students had never sought personal counseling. These findings suggest, although students showed moderate levels of stress, they may have coped well because they had non-professional support systems in place.

Finally, research question seven asked students to rate their level of spirituality and religiosity. In the study conducted by Donahoo and Caffey (2010), to varying degrees, students attributed their successful transitions into college, academic performance, career selection, ability to cope with stress, and the desire to accept and improve the lives of others to their church involvement, religious practice, and spirituality. Walker and Dixon (2002) concluded that the support and encouragement students received from churches motivated them to stay and succeed in school. In the present study, students generally indicated that they sometimes attended a religious service within the past year, and assessed their level of religiousness and spirituality as average or above average. Thus, many of the students were involved in their faith to some degree, which suggests that there may be some correlation to why the participants rated their level of academic ability and performance very highly, and that they may have successfully matriculated through to graduation.

Four hypotheses were presented in this study: (1) Black students will report being dissatisfied with access to campus resources, (2) Black students will report having absent or negative relationships with their instructors, (3) Black students will report having strong feelings of isolation and alienation, and (4) Black students will report being dissatisfied overall with the university as an institution. The hypothesis that Black students would report being dissatisfied

with access to campus was not supported in this study. As presented in chapter two, previous research has shown that many PWIs fail to provide the resources necessary for Black students to excel academically. In the present study, students responded that they were satisfied with the help they had received from tutors, mentors, Financial Aid, and academic advisors, and that the selected university had been successful at preparing them to achieve financial security and their career goals. The university seems to have had effective resources in place to help Black students succeed.

The hypothesis that Black students will report having absent or negative relationships with their instructors was not supported in this study. Although previous research has shown that minority students prefer minority instructors, the participants in the present study indicated that the faculty and staff were appealing and had positive relationships with them, even though the majority of faculty and staff at the selected university were non-minorities. These results suggest that university faculty and staff showed a great amount of support to minority students.

The third and fourth hypotheses, that (3) Black students would feel strongly isolated and alienated, and (4) would report being dissatisfied overall with the institution were not supported in the present study. As presented in chapter two, previous research suggests that many minority students do not succeed at the collegiate level for any of various factors, such as alienation, isolation, racism, discrimination, and intimidation. However, the participants in the present study indicated that they had positive relationships with both faculty and staff, were involved in one or more co-curricular organization(s) on campus, and generally responded that they felt satisfied with the university as an institution. Thus, Black students at the selected institution had, in general, found a sense of belonging on campus, which suggests that students, faculty, and staff were generally welcoming to others, regardless of race.

## Recommendations for Future Research

The researcher received a low response rate and was therefore unable to assess African American perceptions of their experiences at the university on a larger scale. To avoid such a low response rate in subsequent research, future researchers should create more incentive for participation and/or devise other methods of making the instrument accessible to the participants.

In addition, future researchers may want to use a mixed method design including qualitative research. Utilizing qualitative research methods will allow future researchers to further examine participants for more details about their experiences. In addition, in a qualitative study, the researcher is able to clarify questions that may cause confusion; not all participants will interpret questions the same way. In the same manner, the researcher may not interpret participants' responses in the same way that the participants intended. More specifically, researchers may want to ask participants if they would feel more comfortable relating to and building relationships with African American faculty and staff compared to Caucasian faculty and staff.

Finally, future researchers may want to administer their surveys and request that students participate in qualitative studies during the Fall semester, when fewer graduate students have already administered their surveys and made requests to potential participants.

### Conclusion

Overall, African American students at the selected university felt that they were welcome, had support from faculty and staff, were satisfied with the university, in general and its resources, were achieving academically at a high level, demonstrated little to no unhealthy or risky behaviors, were moderately involved in the church, yet had moderate emotional and mental stress. The students were also moderately to highly involved on campus. The participants

indicated that they greatly appreciated the level of diversity and the number of diverse cocurricular activities available to them at the selected institution. Based on the literature in Chapter II and the results of this study, one can conclude that the students who participated in this study remained at the university and matriculated successfully through graduation without major issues that would have lead to attrition. Also, it can be concluded that African American students at the university had more positive perceptions of the university than other African American PWI students had of their institutions.

## References

- Astin, A. W. (1984). Student involvement: A developmental theory for higher education.

  \*Journal of College Student Personnel, 25(4), 297-308.
- Astin, A.W. (1991). Assessment for excellence: The philosophy and practice of assessment and evaluation in higher education. Phoenix, AZ: The American Council on Education Oryx Press Series on Higher Education.
- Astin, A. W. (1993). What matters in college? Four critical years revisited. San Francisco, CA: Jossey-Bass.
- Barksdale, C. L. & Molock, S. D. (2009). Perceived norms and mental health help seeking among African American college students. *Journal of Behavioral Health Services & Research*, 36(3), 285-299.
- Black Student College Graduation Rates Inch Higher But a Large Racial Gap Persists. (2007).

  The Journal of Higher Education. Retrieved from 
  http://www.jbhe.com/preview/winter07preview.html.
- Cabrera, A. F., Nora, A., Terenzini, P. T., Pascarella, E., & Hagedorn, L. S. (1999). Campus racial climate and the adjustment of students to college: A comparison between White students and African American students. *The Journal of Higher Education*, 70, 134-160.
- Cheatham, H. E., Tomlinson, S., & Paterson, A. (1964). Self-concept and school achievement. Sociology of Education, 17, 271-283.
- Cokley, K. O. (2000). An investigation of academic self-concept and its relationship to academic achievement of African American college students. *Journal of Black Psychology*, 26, 148-164.

- Credle, J. O., & Dean, G. J. (1991). A comprehensive model for enhancing black student retention in higher education. *Journal of Multicultural Counseling & Development*, 19(4).
- DeSousa, D.J., & King, P.M. (July 1992). Are white students really more involved in collegiate experiences than black students? *Journal of College Student Development*, 33, 363-369.
- Donahoo, S. & Caffey, R. A. (2010). A sense of home: The impact of church participation on African American college students. *Journal of Research on Christian Education*, 19(1), 79-104.
- Emblen, J. D. (1992). Religion and spirituality defined according to current use in nursing literature. *Journal of Professional Nursing*, 8(1), 41-47.
- Feagin, J. R. (1992). The continuing significance of racism: Discrimination against Black students in White colleges. *Journal of Black Studies*, 22(4), 546-578.
- Fordham, S. (1998). Racelessness as a factor in Black student school success: Programmatic strategy or pyrrhic victory. *Harvard Educational Review*, *58*, 43-84.
- Fortson, S. B. (1997). An evaluation of a program to influence academic self-concept among African American male college students. *Journal of Employment Counseling*, 34(3), 104-107.
- Furr, S. R., & Elling, T. W. (2002). African American students in a predominantly white university: Factors associated with retention. *College Student Journal*, *36*, 188-202.
- Gregory, S. T. (2000). Selected innovations in higher education designed to enhance the racial climate for students of color in predominantly white colleges and universities. Paper presented at the annual meeting of the American Educational Research Association, New Orleans, LA.

- Horn, L., Berger, R., & Carroll, D. C. (2004). National Center for Education Statistics. College persistence on the rise: Changes in 5-year degree completion and post-secondary persistence rates between 1994 and 2000. Washington, D. C.: U.S. Department of Education. Retrieved on September 24, 2011, from http://nces.ed.gov/pubs2005/2005156.pdf.
- Hudson, W. E., Henderson, D., & Henderson, J. (2002). The retention of entering and returning freshmen enrolled at Florida A&M University's School of General Studies during the fall semesters 1997-2001. *Education*, 123(1), 210.
- Hughes, M. S. (1987). Black students' participation in higher education. *Journal of College Student Development*, 28, 532-545.
- Kemp, A. D. (1990). From matriculation to graduation: Focusing beyond minority retention.

  \*Journal of Multicultural Counseling & Development, 18(3), 144-150.
- Lang, M. (1986). Black student retention at predominantly black institutions: Problems issues alternatives. *Western Journal of Black Studies*, 10(2), 48-54.
- Lang, M. (1992). Barriers to Black's education achievement in higher education: A statistical and conceptual review. *Journal of Black Studies*, 22, 510-522.
- Lett, D. F. (2003). Psychological barriers associated with matriculation of African American students in predominantly White institutions. *Journal of Instructional Psychology*, 30(3), 189-196.
- Levin, J. S. (1998). Presidential influences, leadership succession, and multiple interpretations of organizational change. The Review of Higher Education, 21(4), 405-425.

- Lundberg, C. A., & Schreiner, L. A. (2004). Quality and frequency of faculty-student interaction as predictors of learning: an analysis by student race/ethnicity. *Journal of College Student Development*, 45(5), 549-565.
- Majors, R., & Billson, J. B. (1992). Cool pose: The dilemmas of Black manhood in America.

  New York, NY: Touchstone.
- Negga, F., Applewhite, S., & Livingston, I. (2007). African American college students and stress: School racial composition, self-esteem, and social support. *College Student Journal*, 41(4), 823-830.
- Nettles, M. T. (1988). Factors related to Black and White students' college performance. In M. T. Nettles (Ed.), Toward Black undergraduate student equality in American higher education (pp. 18-34). New York: Greenwood Press.
- Oliver, M. L., Rodriguez, C. J., & Mickelson, R. A. (1985). Brown and black in white: The social adjustment and academic performance of Chicano and Black students in a predominantly White university. *Urban Review*, 17, 3-23.
- Parks, S. D. (2000). Big questions, worthy dreams: Mentoring young adults in their search for meaning, purpose, and faith. San Francisco, CA: Jossey-Bass.
- Perna, L. W. (2006). Understanding the relationship between information about college costs and financial aid and students' college-related behaviors. *American Behavioral Scientist*, 49, 1620-1635.
- Polite, V. C., & Davis, J. E. (1999). Introduction: Research focused on African American males.

  In V. C. Polite & J. E. Davis (Eds.), African American males in school and society:

  Practices and policies for effective education, (xiii). New York: Teachers and College

  Press.

- Schwitzer, A. M., Griffin, O. T., Ancis, J. R. & Thomas, C. R. (1999). Social adjustment experiences of African American college students. *Journal of Counseling and Development*, 77(2), 189-204.
- Sedlacek, W. E. (1987). Black students on White campuses: 20 years of research. *The Journal of College Student Personnel*, 484-503.
- Smedley, B. D., Myers, H. F., & Harrell, S.P. (1993). Minority-status stresses and the college adjustment of ethnic minority freshmen. *The Journal of Higher Education*, 64(4), 434-452.
- St. John, E. P., Paulsen, M. B., & Carter, D. F. (2005). Diversity, college, and postsecondary opportunity: An examination of the financial nexus between college choice and persistence for African Americans and Whites. *The Journal of Higher Education*, 76, 553-569.
- The racial gap in college student graduation rates. (2011). *The Journal of Higher Education*. Retrieved from http://www.jbhe.com/2011/10/the-racial-gap-in-college-student-graduation-rates/
- Tinto, V. (1975). Dropout from higher education: A theoretical synthesis of recent research.

  \*Review of Educational Research, 45(1), 89-125.
- Titus, M. A. (2006). Understanding the influence of the financial context of institutions on student persistence at four-year colleges and universities, *The Journal of Higher Education*, 77, 353-375.
- Vital signs: Statistics that measure the state of racial inequality. (2010). *The Journal of Higher Education*, 67. Retrieved from http://www.jbhe.com/vital/index.html.
- Vitell, S. J., Paolillo, G. P., & Singh, J. J. (2005). Religiosity and consumer ethics. *Journal of Business Ethics*, 57(2), 175-181.

Wyman, F. J. (1997). A predictive model of retention rate at regional two-year colleges.

Community College Review, 25(1), 29–58.

## Appendix A

### African American Student Survey

Instructions: Do not include your name or any other identifying information on this form. If you wish to have your name included in the drawing for one of two VISA Gift Cards, please indicate your interest by simply completing the entire survey.

### **Demographic Information**

1.	Gender:			
	1.1. Male			
	1.2. Female			
2.	Your age at the time of completi	ng this survey:	Years	Months
3.	With which of the following raci	ial groups do you	most identify?	
	3.1. African American			
	3.2. Black			
	3.3. Mixed Races			
4.	Please indicate your current enro	ollment status.		
	4.1. Full-time student			
	4.2. Part-time student			
5.	Place indicate your current class	standing:		
	5.1. Freshman (0-29 Credit Hou	rs Earned)		
	5.2. Sophomore (30-59 Credit H	Iours Earned)		
	5.3. Junior (60-89 Credit Hours	•		
	5.4. Senior (90+ Credit Hours E	arned)		
6.	Please indicate your current relig	gious preference:		
	6.1. Baptist	6.2. Buddhis	st	6.3. Church of Christ
	6.4. Eastern Orthodox	6.5. Episcop	alian	6.6. Hindu
	6.7. Islamic	6.8. Jewish		6.9. Lutheran
	6.10. Methodist	6.11. Presbyt		6.12. Quaker
	6.13. Roman Catholic		n Day Adventis	
	6.15.Unitarian/Universalist			st/Congregational
	6.17. Other Christian	6.18. Other R	Religion	
	6.19. No Religious Preference or	Affiliation		
7.	How active are you in the religion	on you indicated in	question numb	per 6?
	7.1. Inactive			

	Place indic	ate which it	tem best reflects	your preferenc	e in selecting thi	s university:
	8.1. First c 8.2. Secon	d choice				
	8.3. Third 8.4. Less the	choice han third ch	oice			
9.	How effect	ive was the	university at reci	ruiting you?		
	9.3. Neithe	what effecti or effective what ineffec	nor ineffective			
10.	Please desc	ribe why yo	ou answered #9 a	bove as you di	id.	
11.		Please use	ns that might have the scale below t			
		are true:			iever at winon et	en of the following
	1		2	3	4	5
Not I	1 nfluential		2 nat Influential			
	nfluential	Somewh		3	4	5
	nfluential	Somewh	at Influential	3	4	5
11.1.	nfluential  My relative	Somewhers wanted m	nat Influential ne to come here	3 Neutral 4	4 Influential	5
11.1.	nfluential  My relative	Somewhers wanted m	nat Influential ne to come here	3 Neutral 4	4 Influential	5
11.1.	nfluential  My relative  1  The univers	Somewhers wanted many 2 sity has a vereceity as a v	nat Influential ne to come here 3 ery good academi	3 Neutral  4 ic reputation  4	4 Influential 5	5
11.1. 11.2.	nfluential  My relative  1  The univers	Somewhers wanted many 2 sity has a vereceity as a v	nat Influential ne to come here 3 ery good academi	3 Neutral  4 ic reputation  4	4 Influential 5	5

7.2. Somewhat Active7.3. Very Active

Recruitment

	1	2	3	4	5
11.5.	The university	y has a lower co	ost of attendan	ce	
	1	2	3	4	5
11.6.	A university a	ndmission coun	selor advised r	ne	
	1	2	3	4	5
11.7.	I wanted to liv	ve near home			
	1	2	3	4	5
11.8.	I was not offe	red aid by my	first college ch	oice	
	1	2	3	4	5
11.9.	I was attracted	d by the religio	us affiliation /	orientation of th	e university
	1	2	3	4	5
11.10.	Information fi	rom the univers	sity's website		
	1	2	3	4	5
11.11.	A visit to the	campus			
	1	2	3	4	5
Matri	culation				
12.	Use the scale statements.	below to identi	fy the level at	which you agree	with the following
	1 Strongly Disa	2 agree Disag	3 ree Neuti	4 ral Agree	5 Strongly Agree
12.1.	Based on the standing.	time I have bee	en attending the	e university, I sh	ould be at a higher class
	1	2	3	4	5
12.2.	During my tir		rsity, I have ha	d no problems e	arning 30 credit hours per

	1	2	3	4	5	
12.3.	I am consider	ing transferring	to a different	institution.		
	1	2	3	4	5	
Reten	tion					
13.	Below are sor Using the sca	me aspects of colle below, please	ollege life at the indicate the	he universit	y that you may ch each item is	find appealing. appealing to you.
	1 Not Appealin	ng Somewha	2 at Appealing	3 Neutral	4 Appealing	5 Very Appealing
13.1.	Difficult subj	ects				
	1	2	3	4	5	
13.2.	Having fun					
	1	2	3	4	5	
13.3.	Having to spe	end lots of time	studying			
	1	2	3	4	5	
13.4.	Learning sub	jects not covere	ed in high scho	ool		
	1	2	3	4	5	
13.5.	Being away f	rom home				
	1	2	3	4	5	
13.6.	Being on you	ır own				
	1	2	3	4	5	
13.7.	Meeting new	and different p	eople			
	1	2	3	4	5	
13.8.	Prepares you	to achieve care	eer goals			
	1	2	3	4	5	
13.9.	Prepares you	to achieve fina	ncial security			

	1	2	3	4	5	
13.10	). The facul	ty and staff				
	1	2	3	4	5	
	Ţ	Jsing the scal	e below, please	e rate your satis	sfaction with th	e item.
Extr	1 emely Nega	tive Neg	2 gative Ne	-	4 sitive Ex	5 stremely Positive
14.	How woul	ld you descril	e your overall	experience at t	he university t	hus far?
	1	2	3	4	5	
15.	On what l	basis have yo	u answered #1	sa as you did?		
Extro	1	-	below, please  2  gative Ne	•	4	5 tremely Positive
16.	How wou		be your relatio	nship with univ	ersity faculty (	professors and class
	1	2	3	4	5	
17.	On what l	oasis have yo	u answered #14	tc as you did?		· 
	U	sing the scale	below, please	rate your satisf	action with eaction	ch item.
	1	-	2		4	5
Extre	emely Nega	tive Neg			sitive Ex	tremely Positive

1	1	2	3	4	5	
. (	On what b	pasis have you a	answered #10	6c as you d	lid?	
. 1	Would yo	u recommend the	he university	to another	r African American?	
	20.1. YE 20.2. NO					
		wered YES to # to consider atte			say that might interes	st another African
- -						
d	during the	e past <u>year</u> .		3	cy at which you have	5
d	during the	e past <u>year</u> .		-		-
ď	during the  1 Never	e past <u>year</u> .	er Som	3	4	5
ď	during the  1 Never  Attended	e past <u>year</u> .  2  Almost Neve	er Som	3	4	5
.1. A	during the  1 Never  Attended	e past <u>year</u> .  2  Almost Neve a religious servi	e <b>r Som</b> ice	3	4 Almost Always	5
.1. A	1 Never  Attended a	e past <u>year</u> .  2  Almost Neve a religious servi	e <b>r Som</b> ice	3	4 Almost Always	5
.1. A 1.2. V	during the  1 Never  Attended :  1 Was bored	e past <u>year</u> .  2 Almost Neve a religious servi	er Som	3 etimes	4 Almost Always	5
.1. A 1.2. V	during the  1 Never  Attended :  1 Was bored  1 Participate	e past <u>year</u> .  2 Almost Neve a religious servi  2 d in class	er Som	3 etimes	4 Almost Always	5
11412V13F1.	during the  1 Never  Attended a  Was bored  1 Participate  1	2 Almost Neve a religious servi 2 d in class 2 ed in organized	or Somo	3 etimes 4	4 Almost Always  5	5
11412V13F1.	during the  1 Never  Attended a  Was bored  Participate  Studied w	e past year.  2 Almost Neve a religious servi 2 d in class 2 ed in organized 2	or Somo	3 etimes 4	4 Almost Always  5	5
1. A. 1. A. 1. 3. F. 1. 4. S. 1.	during the  1 Never  Attended a  Was bored  Participate  Studied w	a religious serving  d in class  2  d in organized  2  cith other studen  2	or Sometice  3  presentation  3	3 etimes 4	4 Almost Always  5  5	5

22.6.	Drank beer				
	1	2	3	4	5
22.7.	Drank wine or	liquor			
	1	2	3	4	5
22.8.	Felt overwhelm	med with all yo	ou had to do		
	1	2	3	4	5
22.9.	Felt depressed				
	1	2	3	4	5
22.10.	Performed vol	unteer work			
	1	2	3	4	5
22.11.	Asked a teach	er for advice af	ter class		
	1	2	3	4	5
22.12.	Voted in a stud	dent election			
	1	2	3	4	5
22.13.	Arrived late to	class			
	1	2	3	4	5
22.14.	Participated in	student govern	nment		
	1	2	3	4	5
22.14.	Joined a frater	nity or sorority	7		
	1	2	3	4	5
22.15.	Played varsity	/intercollegiate	athletics		
	1	2	3	4	5
22.16.	Participated in	student protes	sts or demonstra	ations	
	1	2	3	4	5

	Not Helpful	Slightly Hel	lpful	Neutral	Helpful	Very Helpful
	1	2		3	4	5
23	_	le below, please from each iten		level of help	you have reco	eived during your time at
Attriti	ion					
	1	2	3	4	5	
22.26.	Gotten into a	physical fight /	altercation	on		
	1	2	3	4	5	
22.25.	Been on acad	emic probation				
	1	2	3	4	5	
22.24.	Strengthened	my religious be	eliefs / co	nvictions		
	1	2	3	4	5	
22.23.	Participated in	n student clubs	/ groups			
	1	2	3	4	5	
22.22.	Socialized wi	th someone in a	another ra	icial / ethnic	group	
	1	2	3	4	5	
22.21.	Communicate	ed regularly wit	h my pro	fessors		
	1	2	3	4	5	
22.20.	Sought person	nal counseling				
	1	2	3	4	5	
22.19.	Participated in	n student protes	sts or dem	onstrations		
	1	2	3	4	5	
22.18.	Been satisfied	l with the unive	ersity			
	1	2	3	4	5	
22.17.	Considered tr	ansferring to a	different	college or u	niversity	

	1	2	3	4	5	
23.2.	Tutors					
	1	2	3	4	5	
23.3.	Financial Aid	l .				
	1	2	3	4	5	
23.4.	Academic Ac	lvisors				
	1	2	3	4	5	
23.5.	Faculty					
	1	2	3	4	5	
23.6.	Staff					
	1	2	3	4	5	
24.	Using the sca with the avera yourself.	le below, please age person your	e rate yourself of age. We want	on each of the for the most accura	ollowing tra ate estimate	nits as compared e of how you see
24.	with the avera	le below, please age person your	e rate yourself or age. We want	on each of the for the most accura	ollowing tra	aits as compared e of how you see
	with the avera yourself.	age person your	age. We want	the most accura	ate estimate	e of how you see
	with the avera yourself.	age person your  2  bove Average	age. We want	the most accurate the most acc	ate estimate	e of how you see  5
Highe	with the avera yourself.  1 st 10% A	age person your  2  bove Average	age. We want	the most accurate the most acc	ate estimate	e of how you see  5
<b>Highe</b> : 24.1.	with the avera yourself.  1 st 10% A Academic abi	2  bove Average  lity  2	age. We want  3  Average	the most accura  4  Below Ave	ate estimate	e of how you see  5
<b>Highe</b> : 24.1.	with the average yourself.  1 st 10% A Academic abi	2  bove Average  lity  2	age. We want  3  Average	the most accura  4  Below Ave	ate estimate	e of how you see  5
<b>Highe</b> : 24.1.	with the avera yourself.  1 st 10% A Academic abi  1 Artistic Abilit	2 bove Average dity 2	age. We want  3  Average	the most accurate the most acc	ate estimate erage	e of how you see  5
<b>Highe</b> : 24.1.	with the avera yourself.  1 st 10% A Academic abi  1 Artistic Abilit	2 bove Average dity 2	age. We want  3  Average	the most accurate the most acc	ate estimate erage	e of how you see  5
<b>Highe</b> : 24.1.	with the avera yourself.  1 st 10% A Academic abi  1 Artistic Abilit  1 Computer ski	2 bove Average dity 2 ty 2 dls	age. We want  Average	the most accurate the most acc	ate estimate erage 5	e of how you see  5

23.1. Mentors

24.5.	Creativity				
	1	2	3	4	5
24.6.	Drive to achie	eve			
	1	2	3	4	5
24.7.	Emotional he	alth			
	1	2	3	4	5
24.8.	Leadership at	oility			
	1	2	3	4	5
24.9.	Mathematical	l ability			
	1	2	3	4	5
24.10.	Physical heal	th			
	1	2	3	4	5
24.11.	Public speaki	ng ability			
	1	2	3	4	5
24.12.	Religiousness	S			
	1	2	3	4	5
24.13.	Self confiden	ce (intellectual	)		
	1	2	3	4	5
24.14.	Self understa	nding			
	1	2	3	4	5
24.15.	Spirituality				
	1	2	3	4	5
24.16	Understandir	ng of others			
	1	2	3	4	5

	1	2	3	4	5	
Gradi	uation					
25.	Use the scale statements.	below t	o identify the le	evel at which	you agree wi	th the following
	1 Strongly Dis	sagree	2 Disagree	3 Neutral	4 Agree	5 Strongly Agree
25.1.	I sometimes	feel that	I will not gradu	ate on time.		
	1	2	3	4	5	
25.2.	I feel like dro	opping o	ut of school.			
	1	2	3	4	5	
25.3.	I do not have	enough	financial aid to	continue goi	ng to school.	
	1	2	3	4	5	
25.4.	I will transfer	r to a dif	ferent institutio	n because I de	o not feel wel	come at this university
	1	2	3	4	5	
TT1 1	C . 1 .	.1		• 41 •	1	1:

24.17. Writing ability

Thank you for taking the time to participate in this research project. To send in your responses, simply click on the *Submit* button indicated below.

#### APPENDIX B

Instruments Modified for the African American Student Survey

Instrument 1

The University System of Georgia's
African-American Male Initiative
RESEARCH
EXTERNAL
REPORT
On Attitudes and Barriers
Impacting the Participation
of African-American Males
in the University System of Georgia
Matlock Advertising & Public Relations
and
Paul A. Warner Associates

### USG GRADUATES TELEPHONE RESEARCH QUESTIONNAIRE

Good morning/afternoon/evening, my name is	of Paul A. Warner
Associates,	
Inc. We are conducting a survey of African-American males between	18-25, their parents,
educators in	
high schools and University System of Georgia graduates regarding t about	
higher education. Please be assured that this is only a survey and we anything. I	
know how valuable your time is, so this survey will take no longer th	an fifteen minutes. May I
please	
speak with	
IF REPONDENT IS NOT AVAILABLE, DETERMINE A TIME	
AND DATE WHEN HE WILL BE AT THIS NUMBER. CALL BA	CK
DATE	
TIME AM/PM	
1. To be certain that we are speaking with the right individuals, which	ch of the following describes
you?	
READ LIST	
You are the parent of an African-American male who is in the	
sixth grade or above	
	QUESTIONNAIRE
You are a teacher but not a guidance counselor of	
male students in the sixth grade or above	GO TO "EDUCATORS
You are a guidance counselor of	and GUIDANCE
male students in the sixth grade or above	COUNSELORS"
-	QUESTIONNAIRE

You are an African-American male age 18-25 who is not currently
enrolled in a higher education institution
QUESTIONNAIRE
You are a graduate of the University System of Georgia 5 CONTINUE
None of the above
TERMINATE
6. Which college or university did you attend?
Albany State University
· · · · · · · · · · · · · · · · · · ·
Augusta State University
Coastal Georgia Community College 3
Georgia Perimeter College
Georgia Southern University 5
Georgia State University 6
Macon State College 7
University of Georgia 8
3. Which of the following individuals played a positive role in determining your academic career
path?
READ LIST
684. Which one individual played the most significant role in determining your academic
career path?
ACCEPT ONLY ONE RESPONSE
Q3 Q4
High school principal 1 1
High school teachers
High school athletic coach
High school guidance counselor 4 4
Your parents/guardian 5 5
5. Why do you feel that way about (PERSON MENTIONED IN Q4)?
6a. At what age do you remember first seriously considering what career you wanted?
6b. What career were you considering at Q6a
age?
6c. What career are you currently
·
pursuing?
7. What were the issues that seriously could have gotten in your way of entering college?
The cost
Low grades
Low SAT scores 3
Insufficient college prep courses 4
Other SPECIFY

PROBE						
8b Which, if any, of the issues that seriously could har college are particularly true for African-American ma			n in the v	vay of y	ou entering	
PROBE						
8c. Why do you feel that way?						
9a. Which of the following describes your impression 9b. Which of those impressions about college life was 9c. Which of those impressions about college life was Q Difficult subjects	mo lea 9a 1 2 3 4 5 6 7 8 9	sst a sst aj Q91 2 3 4 5 6 7 8	ppealing? ppealing? Q9c 1 2 3 4 5 6 7 8	READ READ	LIST LIST	be
PROBE  10b. At what grade is the best time to start teaching st college? grade  10c. Why at that grade?  11a. What could be done to increase young African-Acompleting college? READ LIST  11b. What particular assistance do you think should be males  will do well academically in college? READ LIST	.me	rica	n males' i	interest	in attending a	nd

Q11a Q11b
Mentors 1 1
Tutors 2 2
Financial assistance 3
Academic advisors 4 4
Other SPECIFY
11c. While you were in high school, were you familiar with the process for applying for financial
aid to attend college?
Yes 1
No 2
12a. How interesting and appealing is it for an African-American male to attend an institution in
the University System of Georgia? READ LIST
Extremely interesting and appealing 1
Somewhat interesting and appealing
Might or might not be interesting and appealing 3
Somewhat uninteresting and not appealing 4
Extremely uninteresting and not appealing 5
12b. What is most interesting and appealing about attending an institution in the University
System of Georgia for an African-American male? READ LIST
In-state location
Affordable cost
Academic reputation/program
Student body diversity 4
Colleges & universities with all
African-American students
Colleges & universities with high
African-American student presence
HOPE scholarship availability
Athletic programs
Athletic scholarships
7012c. What would it take to improve young African-American males' attitude toward
attending an institution in the University System of Georgia?
PROBE
interesting
and appealing to African-American males?
PROBE
13b. What would be most interesting and appealing to African-American males about that
particular
college or university?
PROBE
13c. What would it take to get more African-American males to attend that college or university?
PROBE

14. Once African-American males are in college what can be done to assure that they will complete
college?
PROBE
15a. How effective are USG colleges and universities at recruiting African-American males? READ LIST
Very effective 1
Somewhat effective
Neither effective nor ineffective3
Somewhat ineffective4
Very ineffective5
15b. Why do you feel that way?
16a. How would you describe your experience at your alma mater? READ LIST
Extremely positive
Very positive 2
Neither positive nor negative 3
Very negative 4
Extremely negative 5
16b. Why do you say that?
17a. Would you recommend your alma mater to another African-American male? Yes 1
No 2
17b. Why do you say that?
18a. Where would you most likely see or hear a message about attending college or going back
to college?
DO NOT READ LIST
7118b. FOR EACH NOT MENTIONED IN Q 18a ASK: Which of the following places would
you be likely
to see or hear a message about attending college or going back to college? READ LIST  Q18a Q18b
Recruitment letters 1 1
TV ads 2 2
Radio ads 3 3
Internet 4 4
Magazine ads 5 5
Newspaper ads 6 6
Billboards
Movie theaters
Mall kiosks 9 9

Bus shelters/ Bus ads 10 10 Other (Specify)
19a. How would you characterize your experience at a USG institution?  Very satisfying
20. Who would you recommend as role models and spokespersons in advertising messages about attending college or going back to college?
THANK AND CONCLUDE  72 MALES 18-25 TELEPHONE RESEARCH QUESTIONNAIRE  Good morning/afternoon/evening, my name is of Paul A. Warner  Associates, Inc. We are conducting a survey of African-American males between 18-25, their parents, educators in high schools and University System of Georgia graduates regarding their attitudes and opinions about higher education. Please be assured that this is only a survey and we are not trying to sell you anything. I know how valuable your time is, so this survey will take no longer than fifteen minutes. May I please speak with IF REPONDENT IS NOT AVAILABLE, DETERMINE A TIME AND DATE WHEN HE WILL BE AT THIS NUMBER. CALL BACK DATE TIME AM/PM  1. To be certain that we are speaking with the right individuals, which of the following describes you?  READ LIST  You are the parent of an African-American male who is in the
rot are the parent of all fine from the sixth grade or above
You are an African-American male age 18-25 who is not currently enrolled in a higher education institution

None of the above
TERMINATE
7. What is your current grade level? (READ LIST)
Less than 6th Grade
6th to 8 <sup>th</sup> Grade
9th to 10 Grade
High school Junior (11th Grade) 4
High school Senior (12th Grade) 5
Graduated high school but not in college6
Did not graduate from high school and not in school7
Attending technical school
Freshman in college
Sophomore in college
Junior or above in college
8. What is the highest level of education your parents or guardians have achieved?
Less than a high school graduate 1
Graduated high school
Attending technical school
Attended college 4
Graduated college 5
Don't know 6
9. Which of the following colleges or universities is closest to where you live? READ LIST
Albany State University 1
Augusta State University 2
Coastal Georgia Community College 3
Georgia Perimeter College 4
Georgia Southern University 5
Georgia State University 6
Macon State College 7
University of Georgia 8
Don't know
7a. Which of the following individuals played a positive role in determining your academic
career path?
READ LIST
7b. Which one individual played the most positive role in determining your academic career
path? READ
LISTOF THOSE MENTIONED IN 7a but ACCEPT ONLY ONE RESPONSE
Q7a Q7b
The school principal 1 1
The teacher
The athletic coach
The guidance counselor
Your parents/guardian
Other (Specify) 6 6 7c. Why do you feel that way about (PERSON MENTIONED IN Q
7b)?
· · · · · · · · · · · · · · · · · · ·

8a. At what age do you remember first seriously discussing what career you wanted?
8b. What career were you considering at _Q8a age?
8c. What career are you currently pursuing?
8d. How likely do you think you will be to achieve Q8c as a career goal? READ
LIST
Extremely likely 1
Somewhat likely 2
Might or might not 3
Somewhat unlikely 4
Extremely unlikely 5
9a. What is the likelihood that you will enroll in a college or university? READ LIST
9b. IF "EXTREMELY", "SOMEWHAT" OR "MIGHT OR MIGHT NOT" IN Q9a ASK: What
is the likelihood that you will enroll in a public college or university in the State of Georgia?
READ LIST
9c. What is the likelihood that you will enroll in a private college or university in the State of
Georgia?
READ LIST
Q 9a Q9b Q9c
Extremely likely 1 1 1
Somewhat likely 2 2 2
Might or might not 3 3 3
Somewhat unlikely 4 4 4 SKIP TO
Extremely unlikely 5 5 5 Q15a
10a. IF "EXTREMELY", "SOMEWHAT" OR "MIGHT OR MIGHT NOT" IN Q9b ASK:
Which college or university in Georgia are you most interested in attending?
10b. What other colleges or universities in Georgia are you interested in attending?
11 100 4 10 4 1-4 16 11 11 11 11 11
11a. What will most determine if you will enter a college or university?
PROBE
TROBE
11b. What will most determine which college or university you enter?
PROBE
12a. What are the issues that could seriously get in the way of you entering any college?
The cost
Low grades
Low SAT scores
Insufficient college prep courses

college?PROBE
13b. Which, if any, of the issues that could seriously get in the way of you entering college at particularly
true for African-American males?
PROBE
14a. Which of the following describes your impression of college life? READ LIST 14b. FOR THOSE MENTIONED IN 14a ASK: Which of those impressions about college life most appealing? READ LIST 14c. FOR THOSE MENTIONED IN 14a BUT NOT MENTIONED IN 14b ASK: Which of those impressions about college life is least appealing? READ LIST
Q14a Q14b Q14c
75Difficult subjects 1 1 1
Having fun
Having to spend lots of time studying 3 3 3
Learning subjects not covered in high school
Being away from home
Being on your own
Meeting new and different people
Prepares you to achieve career goals 8 8 8
Prepares you to achieve financial security 9 9 9
15a How can the chances of an African-American male attending the college of his choice be
most improved?
PROBE
16a. At what grade is the best time to start teaching students about the requirements for attend college? grade 16b. Why at that
grade?
17a. What could be done to increase young African-American males' interest in attending and
completing
college? READ LIST
17b. What particular assistance do you think should be provided in college so that young
AfricanAmerican males will do well academically? READ LIST
Q17a Q17b
Mentors 1 1
Tutors
Financial assistance 3
Academic advisors 4 4
Other SPECIFY
17c. Are you familiar with the process for applying for financial aid to attend college?
Yes 1
No 2

18a. How would you describe your knowledge about the University System of Georgia? That is
the State of Georgia public colleges and universities? READ LIST
Know about it extremely well 1 ASK Q 18b
Know about it somewhat
Know very little about it
Know nothing at all about it 4
18b What do you know about the University System of Georgia?
18c. How interesting and appealing is it for an African-American male to attend an institution in
the University System of Georgia? READ LIST
Extremely interesting and appealing 1
Somewhat interesting and appealing 2
Might or might not be interesting and appealing 3
Somewhat uninteresting and not appealing 4
Extremely uninteresting and not appealing 5
18d. What is most interesting and appealing about attending an institution in the University
System of
Georgia for an African-American male?
In-state location
Affordable cost
Academic reputation/program
Student body diversity 4
Colleges & universities with all
African-American students
Colleges & universities with high
African-American student presence
HOPE scholarship availability
Athletic programs
Athletic scholarships
18e. What would it take to improve young African-American males' attitude toward attending an
institution in the University System of Georgia? PROBE
PROBE
19a. Of all of the colleges and universities in Georgia you know of, which one would be most interesting
and appealing to African-American males?
19b. What would be most interesting and appealing to African-American males about that
particular
college or university?
19c. What would it take to get more African-American males to attend that college or university?
PROBE
20. Once African-American males are in college, what can be done to assure that they will
20. Once Altical-Americal maies are in conege. What can be done to assule that they will

complete

college?
PROBE
21a. How effective are USG colleges and universities at recruiting African-American males?
READ LIST
Very effective 1
Somewhat effective
Neither effective nor ineffective 3
Somewhat ineffective 4
Very ineffective 5
Why do you feel that way?
22a. Where would you most likely see or hear a message about attending college or going back
to college?
DO NOT READ LIST
22b. FOR EACH NOT MENTIONED IN Q 18a ASK: Which of the following places would you
be likely
to see or hear a message about attending college or going back to college? READ LIST Q22a Q22b
Recruitment letters 1 1
TV ads 2 2
Radio ads 3 3
Internet 4 4
Magazine ads 5 5
Newspaper ads 6 6
Billboards
Movie theaters 8 8
Mall kiosks
Bus shelters/ Bus ads 10 10
Other (Specify)
other (Specify)
23. Who would you recommend as role models and spokespersons in advertising messages about attending college or going back to
college?
THANK AND CONCLUDE
HIGH-SCHOOL STUDENTS AND COLLEGE FRESHMEN AND SOPHORMORES
TELEPHONE RESEARCH QUESTIONNAIRE
Good marring/afternoon/exeming my name is
Good morning/afternoon/evening, my name is of Paul A. Warner Associates, Inc. We are conducting a survey among high school juniors and seniors and college
freshmen and sophomores regarding their attitudes and opinions on school and education. Please
be assured that this is only a survey and we are not trying to sell you anything. May I please
speak with IF REPONDENT IS NOT AVAILABLE, DETERMINE A
TIME AND DATE WHEN HE WILL BE AT THIS NUMBER. CALL BACK
DATEAM/PM

1. What is the highest level of education you have achieved? (READ LIST)
Less than a high school Junior (Less than 10th grade) 1 THANK AND TERMINATE
High school Junior (11th Grade) 2
High school Senior (12th Grade) 3
Graduated high school but not in college4
Did not graduate from high school and not in school 5
Attending technical school6
Freshman in college 7
Sophomore in college 8
Junior or above in college
TERMINATE
2. Did you or are you currently attending a public or private high school?
Public school 1
Private school 2
3. What is the highest level of education achieved by your parents or guardian?
Less than a high school graduate 1
Graduated high school 2
Attending technical school
Attended college4
Graduated college 5
Don't know 6
4. Which of the following colleges or universities is closest to where you live? READ LIST
Albany State University 1
Augusta State University 2
Coastal Georgia Community College 3
Georgia Perimeter College 4
Georgia Southern University 5
Georgia State University 6
Macon State College 7
University of Georgia 8
Don't know9
795a. During high school, which of the following individuals plays or played a positive role in
determining
your academic career path? READ LIST
5b. Which one individual plays or played the most positive role in determining your career path.
READ LISTOF THOSE MENTIONED IN 5a but ACCEPT ONLY ONE RESPONSE
Q5a Q5b
The high school principal 1 1
The high school teacher 2 2
The high school athletic coach
The high school guidance counselor 4 4
Your parents/guardian 5 5
6a. At what age do you remember first considering what career you wanted?
6b. What career were you considering at _Q6a age?
6c. What career have you most recently
considered?
LIST

Extremely likely 1
Somewhat likely
Might or might not 3
Somewhat unlikely 4
Extremely unlikely 5
IF NOT CURRENTLY IN COLLEGE CONTINUE WITH Q7a.
IF CURRENTLY IN COLLEGE SKIP TO Q13d
7a. What is the likelihood that you will enroll in a college or university? READ LIST
7b. IF "EXTREMELY", "SOMEWHAT" OR "MIGHT OR MIGHT NOT" IN Q8a ASK: What
is the likelihood that you will enroll in a college or university in the State of Georgia? READ
LIST
Q 7a Q7b
Extremely likely
Somewhat likely
Might or might not
Somewhat unlikely 4 4 SKIP TO
Extremely unlikely 5 5 Q11c 8a. IF "EXTREMELY", "SOMEWHAT" OR "MIGHT OR MIGHT NOT" IN Q7b ASK: Which
or university in Georgia are you most interested in attending?
of university in Georgia are you most interested in attending:
9a. What will most determine if you enter any college?
PROBE
809b What will most determine which college or university you enter?
PROBE
10. What are the issues that could most seriously get in the way of you entering any college?
The cost
Low grades
Insufficient college prep courses 4
Other SPECIFY 4
Outer of ECH 1
11a. What are the ways to get over the issues that could seriously get in the way of you entering college?

PROBE
11b. Which, if any, of the issues that could seriously get in the way of your entering college are
particularly true for African-American males?
PROBE
11c. How can the chances of an African-American male attending the college of his choice be
most
improved?
PROBE
12. At what grade is the best time to start learning about the requirements for attending college?
Grade
13 Why at that
grade?
14a. Which of the following describes your impression of college life? READ LIST
14b. Which of those impressions about college life is most appealing? READ LIST
14c. Which of those impressions about college life is least appealing? READ LIST
Q14a Q14b Q14c
Difficult subjects 1 1 1
Having fun
Having to spend lots of time studying 3 3 3
Learning subjects not covered in high school 4 4 4
Being away from home
Being on your own 6 6 6
Meeting new and different people 7 7 7
Prepares you to achieve career goals
Prepares you to achieve financial security 9 9 9
15a What could be done to increase your interest in attending and completing college? READ
LIST
15b. What particular assistance do you think you will need to do well academically in college?
READ LIST
Q15a Q15b
81Mentors 1 1
Tutors
Financial assistance 3 3
Academic advisors 4 4
Other SPECIFY
15c. Are you familiar with the process for applying for financial aid to attend college?
Yes 1
No 2

16a. How would you describe your knowledge about the University System of Georgia? READ LIST
Know about it extremely well 1 ASK Q 16b
Know about it somewhat 2
Know very little about it 3 SKIP TO Q17
Know nothing at all about it 4
16b What do you know about the University System of
Georgia?
PROBE
16c. How interesting and appealing is it for an African-American male to attend a University
System of Georgia institution? READ LIST
Somewhat interesting and appealing 2
Might or might not be interesting and appealing 3
Somewhat uninteresting and not appealing 4
Extremely uninteresting and not appealing 5
16d. What is most interesting and appealing about attending an institution in the University
System of Georgia for an African-American male? READ LIST
In-state location
Affordable cost
Academic reputation/program 3
Student body diversity 4
Colleges & universities with all
African-American students 5
Colleges & universities with high
African-American student presence
HOPE scholarship availability
Athletic programs
Athletic scholarships
16e. What would it take to improve your attitude toward attending a University System of
Georgia institution?
PROBE
Of all of the colleges and universities in Georgia, which would be most interesting and
appealing to attend?
17b. What is most interesting and appealing about that college or university?
PROBE
17c What would it take to get you to attend that college or university?
PROPE

18. Once you are in c college?	college, what c	an be done to assu	re that you will o	complete
PROBE				
19a. How effective an	re USG college	es and universities	at recruiting Afr	rican-American males?
READ LIST				
Very effective		1		
Somewhat effective.	• • • • • • • • • • • • • • • • •	2		
Neither effective n	or ineffective.	3		
Somewhat ineffect	tive	4		
Very ineffective		5		
19b. Why do you fee	l that way?			
PROBE				
20a. Where would yo	ou most likely	see or hear a mess	age about attendi	ng college or going back
to college?				
DO NOT READ LIS				
		•		ollowing places would you
be likely to see or hea	_	bout attending coll	lege or going bac	k to college?
	Q22a Q22b			
Recruitment letters	1 1			
TV ads	2 2			
Radio ads	3 3			
Internet	4 4			
Magazine ads	5 5			
Newspaper ads				
Billboards	7 7			
Movie theaters	8 8			
Mall kiosks	9 9			
Bus shelters/ Bus ads	10 10			
Other (Specify)				
21 W/h14	1	1 4.1 4	1	. 4
		role models and s	pokespersons in a	advertising messages about
attending college or g college?				
THANK AND CONG	CLUDE		-	
83 EDUCATORS a		CE COUNSELOR	S	
TELEPHONE RESE			~	
				of Paul A. Warner
Associates Inc. We a	are conducting	a survey of Africa	m-American mal	of Paul A. Warner es between 18-25, their
parents educators in	high schools a	nd University Sys	tem of Georgia g	raduates regarding their
				his is only a survey and we
_				so this survey will take no
longer than fifteen m				IF REPONDENT
			DATE WHEN	HE WILL BE AT THIS
NUMBER. CALL BA			TIME	

1. To be certain that we are speaking with the right individuals, which of the following describe you? READ LIST
You are the parent of an African-American male who is in the
sixth grade or above
QUESTIONNAIRE
You are a teacher but not a guidance counselor of
male students in the sixth grade or above
You are a guidance counselor of CONTINUE
male students in the sixth grade or above
mate students in the sixth grade of above
You are an African-American male age 18-25 who is not currently
enrolled in a higher education institution
QUESTIONNAIRE
You are a graduate of the University System of Georgia 5 GO TO "USG
GRADUATE"
QUESTIONNAIRE
QOEDITOTATA
None of the above
TERMINATE
10. What is the current grade level of the male students for which you are a teacher or guidance
counselor? READ LIST
Less than 6th grade
6th to 8th Grade
9th to 10 Grade
High school Junior (11th Grade)4
High school Senior (12th Grade)5
High school graduates and above
Which of the following colleges or universities is closest to where you teach? READ LIST
Albany State University
Augusta State University
Coastal Georgia Community College 3
Georgia Perimeter College 4
Georgia Southern University 5
Georgia State University 6
Macon State College 7
University of Georgia
Don't know9
4a. Which of the following individuals play positive roles in determining your students'
academic career path? READ LIST
4b. Which one individual plays the most significant role in determining your son's academic
• •
career path? ACCEPT ONLY ONE RESPONSE
Q4a Q4b
The school principal 1 1
Teachers 2 2
Teachers
Teachers 2 2

Q4b)?
5. At what grade does your typical African-American male student first seriously discuss what career he wants?
age 6. Which colleges and universities in Georgia are your African-American male students most interested in attending?
7a. What will most determine if your typical African-American male student enters a college or university?
PROBE
7b. What will most determine which college or university your typical African-American male student enters?
PROBE
8. What are the issues that could most likely get in the way of your typical African-American male student entering any college?  The cost
9. What are the ways to get over the issues that could seriously get in the way of your typical AfricanAmerican male student entering college?
PROBE
10. Which of the following describes the typical African-American male student's impression of college?  READ LIST  11a Which of those impressions about college is most appealing to typical African-American
male students? READ LIST  11b Which of those impressions about college is least appealing typical African-American male students?  PEAD LIST
READ LIST  Q10 Q11a Q11b
Difficult subjects

Having to spend lots of time studying 3 3 3
Learning subjects not covered in high school 4 4 4
Being away from home
Being on your own
Meeting new and different people 7 7 7
Prepares you to achieve career goals
Prepares you to achieve financial security 9 9 9
12 How can the chances of an African-American male attending the college of his choice be
most
improved?
PROBE
13a. At what grade is the best time to start teaching students about the requirements for attending
college?
grade
13b. Why at that
grade?
14a. What could be done to increase young African-American males' interest in attending and
completing
college? READ LIST
14b. What particular assistance do you think should be provided in college so that young
AfricanAmerican males will do well academically? READ LIST
Q14a Q14b
Mentors 1 1
86Tutors
Financial assistance 3 3
Academic advisors 4 4
Other SPECIFY
15. Do you think your African-African male students are familiar with the process for applying
for
financial aid to attend college?
Yes 1
No 2
16a. How would you describe your African-African male students' knowledge about the
University System of Georgia? That is, the 34 State of Georgia public colleges and universities?
READ LIST
Know about it extremely well 1 ASK Q 16b
Know about it extremely wen 1 7181 Q 100  Know about it somewhat
Know very little about it
Know nothing at all about it 4
16b What does the typical African-African male student know about the University System of
Georgia?
O001810.
PROBE

17. How interesting and appealing is it for an African-American male to attend an institution in
the University System of Georgia? READ LIST Extremely interesting and appealing
Somewhat interesting and appealing
Might or might not be interesting and appealing 3
Somewhat uninteresting and not appealing 4
Extremely uninteresting and not appealing 5
18a. What is most interesting and appealing about attending an institution in the University
System of Georgia for an African-American male? 1
Affordable cost
Academic reputation/program 3
Student body diversity 4
Colleges & universities with all African-American students
Colleges & universities with high African-American student presence 6
HOPE scholarship availability 7
Athletic programs 8
Athletic scholarships 9
18b. What would it take to improve young African-American males' attitude toward attending an institution in the University System of Georgia?
PROBE_
19a. Of the colleges and universities in Georgia you know about, which would be most interesting and
appealing to African-American males?
19b. What would be most interesting and appealing to African-American males about those colleges or universities?
PROBE
19c. What would it take to get more African-American males to attend that college or university?
PROBE
20. Once African-American males are in college what can be done to assure that they will
complete
college?
PROBE
21a. How effective are USG colleges and universities at recruiting African-American males? READ LIST
Very effective

Somewhat effective
22a. Where would you most likely see or hear a message about attending college or going back to college?  DO NOT READ LIST  FOR EACH NOT MENTIONED IN Q 18a ASK: Which of the following places would you be
likely to see or hear a message about attending college or going back to college?
Q22a Q22b
Recruitment letters 1 1
TV ads
Radio ads
Internet
Magazine ads
Newspaper ads 6 6 Billboards
Movie theaters
Mall kiosks
Bus shelters/ Bus ads 10 10
Other (Specify)
23. Who would you recommend be used as role models and spokespersons in advertising messages about attending college or going back to college?

#### II. Taken From:

### FACTORS INFLUENCING THE COLLEGE CHOICE OF AFRICAN-AMERICAN STUDENTS ADMITTED TO THE COLLEGE OF AGRICULTURE, FOOD AND NATURAL RESOURCES

A Thesis presented to the Faculty of the Graduate School University of Missouri-Columbia

In Partial Fulfillment
of the Requirements for the Degree
Master of Science

By MARVIN J. BURNS

Dr. Bryan L. Garton, Thesis Supervisor

**AUGUST 2006** 

1.Listed below are several ways you might have learned about the College of Agriculture, Food and Natural Resources (CAFNR). For each source of information, please circle Yes or No if you used the source of information in the college selection process. For each		Did you use the information source? (Circle One)		If you marked "Yes," how useful was the source of information? (Circle One)					
source you mark Yes, please circle how useful that source was in making your decision.  Use the scale 5=Very Useful to 1=Not Useful									
Source of Information	Yes	No	Very	Useful	1	Not	Useful		
Personal conversation with a professor	Y	N	5	4	3	2	1		
b. Personal conversation with a CAFNR representative	Y	N	5	4	3	2_	1		
c. Personal conversation with an MU admissions representative	Y	N	5	4	3	2	*		
d. Letter and/or information mailed from a professor	Y	N	5	4	3	2	1		
e. Letter and/or information mailed from a CAFNR representative	Y	N	5	4	3	2	1		
f. Letter and/or information mailed from an MU admissions representative	Y	N	5	4	3	2	i _		
g. Visit to campus	Y	N	5	4	3	2	1		
h. MU information on a website	Y	N	5	4	3	2			
i. CAFNR information on a website	Y	N	5	4	3	2	ş;		
j. Degree Program (major) information on a website	Y	N	5	4	3	2	, ;		
k. Printed MU publications (brochures, etc.)	Y	N	5	4	3	2	¥		
Visits by MU representative to your school	Y	N	5	<u> </u>	3	2			
m. College comparison guides (Barron's, Peterson's, U.S. News & World Report, etc.)	Y	N	5	4	3	2	Fernit.		
n. TV, radio, newspaper, or magazine advertisements	Y	N	5	4	3	2	1		
<ul> <li>Participation in an on-campus recruitment program (Target Hope, Preview Mizzou, Black and Gold Day, etc.)</li> </ul>	Y	N	5	4	3	2	\$cord		
p. Participation in student activity events on campus (Music, Theatre, FFA, 4-H, etc.)	Y	N	5	4	3	2	1		
<ul> <li>Participation in athletic events on campus (sports camps, state championships, etc.)</li> </ul>	Y	N	5	4	3	2	Ferrit		
. MU information from High School counselor	Y	N	5	4	3	2	1		
. CAFNR information from a High School counselor	Y	N	5	4	3	2	)		

2. How influential were the following factors when making your decision regarding which university to attend?		Level of Influence						
					Not			
		Influential Influentia						
Academic reputation of the university	5	4	3	2	1			
b. Prestige of the university	5	4	3	2	1			
c. Quality of facilities	5	4	3	2	1			
d. Quality and reputation of the faculty	5	4	3	2	1			
e. Quality and reputation of the students	5	4	3	2	1			
f. Preparation for employment	5	4	3	2	1			
g. Cost (tuition, room and board)	5	4	3	2	1			
h. Scholarships awarded	5	4	3	2	1			
i. Availability of other financial aid	5	4	3	2	1			
j. Size of classes	5	4	3	2	1			
k. Campus safety and security	5	4	3	2	1			
l. Variety of majors offered	5	4	3	2	1			
m. Competitiveness of admissions standards	5	4	3	2	1			
n. City in which campus is located	5	4	3	2	1			
o. Distance from home	5	4	3	.2	1			
p. Prominence of university athletic teams	5	4	3	2	1			
q. Opportunities after graduation	5	4	3	2	1			
r. Diversity of the faculty	5	4	3	2	1			
s. Diversity of the students	5	4	3	2	1			
t. Student support services for minorities	5	4	3	2	ı			
u. Quality and reputation of courses	5	4	3	2	1			
v. Career opportunities available for graduates	5	4	3	2	1			

3. How influential was input from t	1550 AUGUS	,		Level	of Infl	uenc	;e
following individuals in your decision to attend the University	1.58 (c. 1) Toyo (d. 1)	ту				Not	
Missouri?	1.8 de 1 - 19.99 M	lueni	tial		Influe	ential	. Not
							Applicable
a. Friend in high school		5	4	3	2		NA
b. Friend in college		5	4	3	2	· ·	NA
c. Parent or guardian		5	4	3	2	******	NA
d. Relative who attended the University of Misson	ıri g	5	4	3	2	• and	NA
e. High school guidance counselor		5	4	3	2		NA
f. High school agriculture teacher		5	4	3	2	•	NA
g. High school science teacher		5	4	3	2	•••••	NA
h. Other high school teacher		5	4	3	.2	*	NA
i. Extension youth specialist	5	5	4	3	2	g.cocod,	NA
. University of Missouri graduate	-	5	4	3	2	*****	NA
k. Graduate of CAFNR		5	4	3	2	*****	NA
l. College recruiter		5	4	3	2	1	NA
m. Relative who attended college		5	4	3	2	1	NA
n. An acquaintance or friend who attended MU		5	4	3	2	1	NA
o. Current CAFNR student		5	4	3	2	1	NA

4. As you think about opportunities for social	Level of Influence							
- [통령 역 교육전기 전통표점점점		rry t						
		Influential Influential						
Quality and availability of recreational services	-5	4	3	2	1			
Quality and availability of student organizations	٢		3	2				
Quality and availability of off-campus activities	5	4	3	2				
. Quality and availability of agricultural competitive teams	5	4	3	2	1			
e. Leisure activities	5	4	3	2	1004			
f. Fraternity and sorority life	5	4	3	2	****			
g. Campus residence halls	5	4	3	2	1			
h. Diversity of student body	5	4	3	2	*			
i. Diversity of ideas on campus	5	4	3	2	1			
k. Quality of social integration	5	4	3	2				

	When did y Please chec		cision mal	cing proce	ess to select a col	lege (	of univ	ersity?
	Before 9 <sup>t</sup> During 1	<sup>th</sup> grade 1 <sup>th</sup> grade			9th grade 12 <sup>th</sup> grade	70,000	□ Du	ring 10 <sup>th</sup> grade
	Vhen did yo Please chec		decision t	o attend t	he University of	Misso	ouri?	
	Before 9 <sup>th</sup> During 1 During 1	<sup>th</sup> grade 0 <sup>th</sup> grade 1 <sup>th</sup> grade		During	12th grade 1st half of 12th 2nd half of 12th		$\Box 2^{nd}$	Yr after 12th Yr after 12th I undecided
	hen die yo Please chec	u finalize your	decision o	f a major	?			
	Before 9 <sup>t</sup> During 1 12th	<sup>th</sup> grade 0 <sup>th</sup> grade			12th grade 1st half of 12th			1 <sup>st</sup> Yr after 12th □ 2 <sup>nd</sup> Yr after
	During 1	1 <sup>th</sup> grade		During	2 <sup>nd</sup> half of 12th			Still undecided
8. To	how many	other universit	ies did yo	u apply? (	(Please check)			
	None	☐ One	□ Two	00000	☐ Three	□F	our	☐ Five or more
9. To	how many	other universit	ies were y	ou admit	ted? (Please chec	k)		
	None	□ One	□ Two		☐ Three	□F	our	☐ Five or more
10.	To which o	other institution	s did you a	apply? (P	lease check all th	at app	oly)	
	Washingto	Iniversity ate University n University						
	Other	Piezse Name						
o	ther	Please Name						
□ o	ther	Please Name						
□ o	ther	Pietase Name						
□ o	ther	11.22.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2	·····					

The CIRP Freshman Survey was obtained from the Higher Education Research Institute, home of the Cooperative Institutional Research Program, and this version was administered in 2005 \_\_\_\_\_\_\_

FIRST	N	LAST			maninger and	When w	ere you born	£?
ME:								
PRESS:						<b>Month</b> (01-12)	Day Ye (01-31)	eər
r:		STATE:	ZJP:	Pi	HONE: -			N.
See as talk control from the control fro	200	5 STUDENT	INFORMATION	FORM		***************************************	terrentikeren en e	gê esero-o-o
MARKING DIRECTION		rom what high :	school did you gra	duate?	16. Where do yo		luring the fal	ii
ur responses will be read b irk reader. Please.	y an optical					advention:	satistic :	
rk reader. Please, se a pencil or black or blue p	Nan	e of High Schoo	)i	·····	With my family or other relatives			
se a perior or black or blue p ii in the oval completely.	7 <b>4</b> 71.			***************************************	College residence hall			
rase cleanly any answer you		October 1			Fraternity or scronty house			
range or "X" out mark if in <b>p</b> e	n. Cny	City State O				s student housir		
ORRECT MARK INCORREC	rom what kind o	on mia king or myr action on you		Other	Other			
and the second s		raduate? (Mark	one)	•	adilities of the state of the s			
PLEASE PROVIDE YOUR Mark here ID NUMBER (as instructed) If directed		<ul> <li>Public school (<u>not</u> charter or magnet)</li> </ul>		ret)	17. Is this colleg		one)	
IO NUMBER (35 IIISHUCIEU)	GROUP GROUP CODE CODE	Public charter	school		First chaice?	94400	s than third	
		Public magnet	school		Second choic		dice?	er.
099000000	3 1 1 2 1 1 1 1	Private religios	zs/parochiał school		Third choice?			
000000000	D 0 0 0 0	Private indepe	ndent college-prep	school				
0000000000	00000	Home school			18. To how man	colleges other	er than this o	MI E
	0 0 0 0 1 10 v	Shat were valir i	cores on the SAT	1	did you app	y for admissio	n this year?	
	ocoo .	nd/or ACT?		•	None 🔘 1	0 40	7-10	
D00000000	ps os	AT VERBAL			2	C 5 C	11 or more	Maria Service
	PROBER 1			***************************************	3	O 6 C		
DODOSTOS	baba   :	IAT MATH			Per district			
		CT Composite .			19. Have you ha	d, or do you le utoring or rem		
			*******	own.		utoring or rem Howing subjec		3
omenigen men de de la constitució de conferencia de la conferencia de la conferencia de la conferencia de la c	11. (	itizenship statu	<b>3</b> .		(Mark <u>all</u> that			(di)
I. Your sex: 🗀 Male 🔝	Female	U.S. citizen			- Andrews			**** 890
2. How old will you be on Dece	mhar 31	⊇ Permanent rei	sdent (green card)		English	************	.0	

of this year? (Mark <u>one</u> )	1	□ Neither			ng	
16 or younger . 10 21	1-24	12. Which of th	ne following statements applies	Mathe	matos	😄 💌
17	5-29	to you? (N		Social	Studies	
	)-39 💢 💮	Lwas bom i	n the United States	Science	ce	·
19 🗇 40	)-54	i came to th	e United States:	Foreig	n Language	
	5 or older . C	Before ag	e6 O	Writin	•	(C) •
***		•	ages 6-12	1	•	
3. Is English your native lange	nana?		12			2 7y <b>.</b>
S. 13 Cilyaan you name lang.	<b></b>	Marie Mar	** · · · · · · · · · · · · · · · · · ·		is the highest academic	
		13. If you or yo	our parents were not born in the	degre obtair	e that you intend to	
4. In what year did you gradua	ate from	United Stat	tes, indicate the country of birth:		one in each column)	, <i>i</i>
high school? (Mark <u>one)</u>					***************************************	# # <b>*</b> -
2005 Old not gra	vduate but	***************************************				
1 8	3.E.D. test. 🔘	You		1	onal certificate	
2003 Never com	4	***************************************		1	iate (A.A. or equivalent)	
2002 orewales. 🗀 high sch	001			1	tor's degree (B.A., B.S., etc.).	
5. Are you enrolled (or enrollis	ng) as a:	Mother		1	rs cegree (M.A., M.S., etc.)	
(Mark <u>one)</u> Full-time st	lucent?				or Ed.D	
Part-time s	Aucent? 🔘		**************************************	M.D.,	D.O., D.D.S., or D.V.M.	OO =
6. How many miles is this col	leae from	Father		J.D. (I	.aw)	OO =
your permanent home? (M		14. Prior to this	s term, have you ever taken	8.D. o	r M DIV. (Divinity)	
5 or less 🔘 11-50 🗇	101-500		r credit at this institution?	Other		00 -
6-10 O 51-100 O	Over 500	_ Yes	○ <b>N</b> o			
7. What was your average gra	da in biah			21. Are yo	our parents: (Mark <u>one)</u>	
school? (Mark one)	ide in ingli	15. Since leavi	ing high school, have you ever	1 7	nive and living with each oth	ver?○ ■
A Or A+ O 6 O	col	taken cour	ses, whether for credit or not for	Both a	dive, divorced or living apart	eo •
A-0 6-0	o a		ny other institution (university,	3	r both deceased?	
8+ () C+ ()	***	4- or 2-year or busines:	r college, technical, vocational,			
<b>5</b> 4 ( <b>5</b> 4 (	4	C Yes	~ ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~			
		100		1		
=== 22. How much of your fit			26. For the activities below, indicate ones you did during the past ye		29. In deciding to go to coll	lege, how \$
22. How much of your fit (room, board, tuttou cover from each of below? [Mark one a each possible source	n, and fees) do yo the sources listed	ou expect to	ones you did during the <u>past ve</u> engaged in an activity frequenti in it you engaged in an activity	ar. If you y, mark r one or	29. In deciding to go to col important to you was ei the following reasons? (Mark <u>one</u> answer for ear possible reason)	lege, how Factor of The State o
(room, board, tuition cover from each of below? (Mark one a	n, and fees) do yo the sources listed	ou expect to	ones you did during the <u>past ve</u> engaged in an activity frequenti in it you engaged in an activity	ar. If you y, mark r one or	Important to you was ex the following reasons? (Mark one answer for ear	lege, how section to section the section that section the
(room, board, tuition cover from each of the below?   Mark one a each possible source	n, and fees) do yo the sources listed mawer for	ou expect to	ones you did during the <u>past ve</u> engaged in an activity frequenti in it you engaged in an activity	ar. If you y, mark r one or	Important to you was ex the following reasons? (Mark one answer for ear	ach of
(room, board, tuition cover from each of below? (Mark one a	n, and fees) do yo the sources listed mawer for	ou expect to	ones you did during the <u>past ve</u> engaged in an activity frequenti in it you engaged in an activity	ar. If you y, mark r one or	Important to you was ex the following reasons? (Mark one answer for ear possible reason)	gc. (D) (D)
(room, board, tuition cover from each of the below? (Mark one a each possible source Family resources (paretatives, spouse, e	n, and fees) do yo the sources listed mawer for the sources listed maker for the sources listed arents	ou expect to	ones you did during the <u>past ve</u> engaged in an activity frequenti ͡ː . If you engaged in an activity	ar. If you y, mark y one or mark () all)	Important to you was exthe following reasons? (Mark <u>one</u> answer for earpossible reason)  My parents wanted me to	ge. (DQ 5) (D(B) 5)
(room, board, tuition cover from each of the below?   Mark one a each possible source   Family resources (pa	n, and fees) do yo the sources listed hiswer for liverants	expect to	ones you did during the <u>past ve</u> engaged in an activity frequent!  If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at if you have not performed the activity during the past year. (Mark <u>one</u> for each item)	at. If you y, mark y one or mark () all)	Important to you was extended following reasons? (Mark <u>one</u> answer for ear possible reason)  My parents wanted me to I could not find a job	gc. (I) (I) (I) (D) (I) (I) ome (D) (I) (I)
(room, board, tuition cover from each of the below? (Mark one a each possible source (parelatives, spouse, each your resources (sawork, work, study, of	n, and fees) do yo the sources listed inswer for  ivents sto) svings from ther income)	expect to	ones you did during the past veengaged in an activity frequentle. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at all you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in croanized	at. If you y, mark y one or nark (a) all)	Important to you was extent following reasons? (Mark one answer for ear possible reason)  My parents wanted me to a could not find a job. Wanted to get away from to To be able to get a better. To gain a general educati	ge. (\$\cap \) (\$\sigma\$)  arms (\$\cap \) (\$\cap \)  job. (\$\tau \) (\$\cap \)
(room, board, tuttion cover from each of t below? [Mark one a each possible source Family resources (pa relatives, spouse, e My own resources (sa work, work-study, of Aid which need not b (orants, scholarship	n, and fees) do yo the sources listed chaser for () vents strongs from the income) () in repaid on military	expect to	ones you did during the past veengaged in an activity frequents. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at all you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.	ar. If you y, mark y, mark y one or mark (1) (1) (2) (2) (3) (3) (4) (4) (4) (4) (4) (4) (4) (4) (4) (4	Important to you was extended following reasons? (Mark one answer for ear possible reason)  My parents wanted me to a could not find a job	ge. (\$\cap \) (\$\sigma\$)  arms (\$\cap \) (\$\cap \)  job. (\$\tau \) (\$\cap \)
(room, board, tuition cover from each of to below? (Mark one a each possible source Family resources (pa retatives, spouse, e My own resources (sa work, work-study, of Aid which need not b (grants, scholarship funding, etc.).	n, and fees) do yo the sources listed inswer for  in and fees) in and fees) in and fees in any fees in and fees in any fees in any fees in any fees in	expect to	ones you did during the past veengaged in an activity frequently. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at all you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.	at. If you y, mark y one or mark of the state of the stat	Important to you was extent following reasons? (Mark one answer for ear possible reason)  My parents wanted me to a could not find a job. Wanted to get away from to To be able to get a better. To gain a general educati	ge . (C (2 (5 )) (E (5 )) .
(room, board, tuition cover from each of to below? (Mark one a each possible source Family resources (pa retatives, spouse, e My own resources (sa work, work-study, of Aid which need not b (grants, scholarship funding, etc.).	n, and fees) do yo the sources listed inswer for  in and fees) in and fees) in and fees in any fees in and fees in any fees in any fees in any fees in	expect to	ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at all you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.	at. If you y, mark y, mark y, mark y one or mark (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job.  Wanted to get away from b. To be able to get a better. To gain a general education and appreciation of idea. There was nothing better to make me a more culture.	ge. C.C.S.S.S.S.S.S.S.S.S.S.S.S.S.S.S.S.S.
(room, board, tuition cover from each of the below? (Mark one a each possible source (parelatives, spouse, each your resources (saviork, work-study, of Aid which need not be (grants, scholarship funding, etc.).  Aid which must be regions, etc.)	n, and fees) do yo the sources listed inswer for the rents, seto).  avings from ther income).  It comes the repaid to military.		ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.	ar. If you y, mark y one or nark ( ) and ( ) a	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job.  Wanted to get away from bit to gain a general educate and appreciation of idea. There was nothing better I to make me a more cultuperson.	ge CO Some CO
(room, board, tuition cover from each of to below? (Mark one a each possible source  Family resources (pa retatives, spouse, e My own resources (sa work, work-study, of Aid which need not b (grants, scholarship funding, etc.).	n, and fees) do yo the sources listed inswer for the rents, seto).  avings from ther income).  It comes the repaid to military.		ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a feacher's nome.	ar. If you y, mark y one or nark ( ) and ( ) a	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job.  Wanted to get away from b. To be able to get a better. To gain a general education and appreciation of idea. There was nothing better to make me a more culture.	ge CO Some CO
(room, board, tuition cover from each of the below? (Mark one a each possible source (parelatives, spouse, each your resources (saviork, work-study, of Aid which need not be (grants, scholarship funding, etc.).  Aid which must be regions, etc.)	n, and fees) do yo the sources listed inswer for the rents, seto).  avings from ther income).  It comes the repaid to military.		ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark  (Not at all you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service. Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a feacher's home.  Smoked organized.	at. If you y, mark y one or mark of the state of the stat	Important to you was extent following reasons? (Mark one answer for ear possible reason)  My parents wanted me to a could not find a job	ge CO S
(room, board, tuttion cover from each of the below? (Mark one a each possible source (paretaives, spouse, each yown resources (savork, work-study, of Aid which need not be (grants, scholarship funding etc.).  Aid which must be regions, etc.)  Other than above.	n, and fees) do yo the sources listed inswer for the rents sto.  sto.  in rents sto.  in repaid as, military  paid	arents' total	ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service. Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a feacher's nome.  Smoked organized.	at. If you y, mark y one or nark of the state of the stat	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b. To be able to get a better. To gain a general education of ide: There was nothing better to make me a more cultuperson.  To be able to make more missions?	ge CO S
(room, board, tultion cover from each of the below? (Mark one a each possible source (paretaives, spouse, each yown resources (sawork, work-study, of Aid which need not be (grants, scholarship funding etc.).  Aid which must be regioned as the control of the con	n, and fees) do yo the sources listed maker for  in the sources listed maker for  stimate of your pa consider income for	arents' total	ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service. Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a feacher's home.  Smoked organized.  Drank beer.  Drank were or liquor.  Felt overwhelmed by all 1 had to do.	at. If you y, mark y one or nark of the state of the stat	Important to you was esthe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job.  Wanted to get away from bit to be able to get a better. To gain a general education and appreciation of idea and appreciation of idea. There was riothing better to make me a more cultiperson.  To be able to make store make interest me.  To prepare myself for grading the control of the prepare myself for grading the control of the c	ge CO S
(room, board, tuition cover from each of the below? (Mark one a each possible source (paretaives, spouse, each work work-study, of Aid which need not be (grants, acholarship funding etc.)	n, and fees) do yo the sources listed maker for () () () () () () () () () () () () ()	expect to	ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a milgious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a teacher's home.  Smoked organized.  Drank beer.  Drank wine or liquor.  Felt overwhelmed by all 1 had to do.  Felt depressed.	ar. If you y, mark y, mark y, mark y one or mark of the state of the s	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b. To be able to get a better. To gain a general education and appreciation of ide. There was nothing better I to make me a more cultuperson.  To be able to make more mit to learn more about thing that interest me.  To prepare myself for grat or professional school.	ge CO S
(room, board, tuttion cover from each of the below? (Mark one a each possible source (paretalives, spouse, each possible sources, spouse, each work work-study, of Aid which need not be (grants, acholarship funding etc.).  Aid which must be regions, each of the dans, etc.).  Other than above.  23. What is your best end income last year? Coources before taxes.	n, and fees) do yo the sources listed maker for () () () () () () () () () () () () ()	expect to	ones you did during the past veengaged in an activity frequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a milipious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a teacher's nome.  Smoked organized.  Drank wine or liquor.  Felt overwhelmed by all 1 had to do.  Felt depressed.  Performed volunteer work.	ar. If you y, mark y one or mark of the state of the stat	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job	ge C C C C C C C C C C C C C C C C C C C
(room, board, tuttion cover from each of the below? (Mark one a each possible source (paretaives, spouse, each work, work-study, of Aid which need not be (grants, scholarship funding, etc.).  Aid which must be regions, etc.).  Other than above.  23. What is your best eventually each courses before taxe.  Less than \$10.00.	n, and fees) do yo the sources listed inswer for it irents sto) wrents ther income) ire repaid as, military irents stimate of your pa consider income fi is. (Mark ane) it is \$50,000 \$80,000	arents' total rom all	ones you did during the past veengaged in an activity irequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Vias a guest in a teacher's home.  Smoked organeties.  Drank beer.  Drank were or liquor.  Felt depressed.  Performed volunteer work.  Played a musical instrument.	ar. If you y, mark y one or mark of the state of the stat	Important to you was esthe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b. To be able to get a better. To gain a general education and appreciation of ide. There was nothing better to make me a more cultuperson.  To be able to make more mit to learn more about thing that interest me.  To prepare myself for grat or professional school. A memor/role model encouraged me to go	go. C.S. S.
(room, board, tuttion cover from each of the below? (Mark one a each possible source (paretaives, spouse, each possible sources (savinces, spouse, each work, work-study, of Aid which need not be (grants, scholarship funding etc.).  Aid which must be regions, etc.)  Other than above.  23. What is your best evincome last year? Cosources before taxe  Less than \$10.00  \$15,000-14,999	n, and fees) do yo the sources listed inswer for it it irents sto.) iverents iter income) iverepaid inswer for iter income) iverepaid inswer for iverepaid inswer for iverepaid inswer for iverents inswer for	expect to	ones you did during the past veengaged in an activity frequenth.  If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a teacher's home.  Smoked organized one.  Crank beer.  Drank were or liquor.  Fell overwhelmed by all I had to do Felt depressed.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice.	ar. If you y, mark y, mark y, mark y and y	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b. To be able to get a better. To gain a general education and appreciation of ide: There was nothing better I. To make me a more cultiperson.  To be able to make more micro team more about thing that interest me.  To prepare myself for grat or professional school. A mentor/role model encouraged me to go.	ge CO Some CO Some CO Some CO SO Some CO SO Some CO SO Some CO SO
(room, board, tutton cover from each of the below? (Mark one a sach possible source (parelatives, spouse, each possible sources (sach possible sources) (prants, scholarship funding etc.)  Aid which must be regionally each possible for than above.  23. What is your best ending each pources before taxes (income last year? Cosources before taxes) (assible of the sources) (ass	n, and fees) do yo the sources listed inswer for it it irents sto.) strents iter income) ine repaid as, military inpaid  stimate of your pa consider income fi is it	arents' total rom all -59,999 9-99,999 9-149,999	ones you did during the past veengaged in an activity irrequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in crash item)  Attended a religious service.  Was bored in crash item.  Tutored another student.  Studied with other students.  Was a guest in a teacher's home.  Smoked orgerettes.  Drank beer.  Drank wine or liquor.  Fell overwhelmed by all I had to do Felt depressed.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice after class.	ar. If you y, mark y, mark y, mark y and y	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job.  Wanted to get away from b To be able to get a better. To gain a general educate and appreciation of ide: There was nothing better to make me a more outlingerson.  To be able to make more mistore mistore mistore me.  To prepare myself for grat or professional school. A mentorhole model encouraged me to go.  To get training for a specicareer.	ge CO Some CO
(room, board, tuttion cover from each of the below? (Mark one a sach possible source (parelatives, spouse, each possible sources (sach possible sources).  Family resources (parelatives, spouse, each your resources (sach work, study, of Aid which need not be (grants, scholarship funding etc.).  Aid which must be regioned as the following etc.).  Other than above.  23. What is your best ending each year? Cosources before taxe.  Less than \$10.00  \$15.000-19.999  \$25.000-24.999  \$25.000-29.999	n, and fees) do yo the sources listed inswer for it it irents sto.) strents story irents story i	arents' total rom all -59,999 9-99,999 9-149,999 9-199,999	ones you did during the past ve engaged in an activity irrequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service. Was bored in drass.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a feacher's nome.  Smoked organities.  Drank beer.  Drank with or fliquor.  Felt depressed.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice after class.  Voted in a student election.	ar. If you y, mark y, mark y, mark y and y	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b. To be able to get a better. To gain a general education and appreciation of ide: There was nothing better I. To make me a more cultiperson.  To be able to make more micro team more about thing that interest me.  To prepare myself for grat or professional school. A mentor/role model encouraged me to go.	ge CO Some CO
(room, board, tuttion cover from each of the below? (Mark one a sach possible source (parelatives, spouse, each possible sources (sach possible sources) (prants, scholarship funding etc.)  Aid which must be refloans, etc.)  Other than above.  23. What is your best endources before taxes  Less than \$10,00  \$15,000-14,999  \$15,000-19,999  \$25,000-24,999  \$25,000-29,999  \$30,000-39,999	n, and fees) do yo the sources listed inswer for it it irents sito.) sitory wents from ther income) is repaid as, military ipaid  stimate of your pa consider income fi as. (Mark one) S80,000 S80,000 S75,000 S100,000 S150,000 S100,000 S200,000	arents' total rom all -59,999 0-149,999 0-249,999	ones you did during the past veengaged in an activity irrequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a teacher's home.  Smoked organized one.  Drank whee or liquor.  Fell overwhelmed by all I had to do Felt depressed.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice after class.  Voted in a student election.  Socialized with someone of	ar. If you y, mark y, one or mark y, mark y, one or mark w. of the second secon	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job.  Wanted to get away from b To be able to get a better. To gain a general educate and appreciation of ide: There was nothing better to make me a more outlingerson.  To be able to make more mistore mistore mistore me.  To prepare myself for grat or professional school. A mentorhole model encouraged me to go.  To get training for a specicareer.	ge CO Some CO
(room, board, tuttion cover from each of the below? (Mark one a sach possible source (parelatives, spouse, each possible sources (sach possible sources).  Family resources (parelatives, spouse, each your resources (sach work, study, of Aid which need not be (grants, scholarship funding etc.).  Aid which must be regioned as the following etc.).  Other than above.  23. What is your best ending each year? Cosources before taxe.  Less than \$10.00  \$15.000-19.999  \$25.000-24.999  \$25.000-29.999	n, and fees) do yo the sources listed inswer for it it irents sto.) strents story irents story i	arents' total rom all -59,999 0-149,999 0-249,999	ones you did during the past veengaged in an activity irrequentity. If you engaged in an activity (Occasionally), Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Vias a guest in a teacher's home.  Smoked organized to break organized organized in a teacher's home.  Peti overwhelmed by all thad to do Felt depressed.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice after class.  Voted in a student election.  Socialized with someone of another racial/ethnic group.	at. If you y, mark y, one or mark y, one or mark w.	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b To be able to get a better. To gain a general education and appreciation of idea and appreciation of idea. There was nothing better to make me a more cultuperson. To be able to make more material interest me. To prepare myself for gracing professional school. A mentor/role model encouraged me to go. To get training for a specioarser. To find my purpose in life.	go. C.
(room, board, tuttion cover from each of the below? (Mark one a sach possible source (parelatives, spouse, each possible sources (sach possible sources) (prants, scholarship funding etc.)  Aid which must be refloans, etc.)  Other than above.  23. What is your best endources before taxes  Less than \$10,00  \$15,000-14,999  \$15,000-19,999  \$25,000-24,999  \$25,000-29,999  \$30,000-39,999	n, and fees) do yo the sources listed inswer for it it irents sito.) sitory wents from ther income) is repaid as, military ipaid  stimate of your pa consider income fi as. (Mark one) S80,000 S80,000 S75,000 S100,000 S150,000 S100,000 S200,000	arents' total rom all -59,999 -99,999 0.149,999 0.0 or more	ones you did during the past ve engaged in an activity irrequentity. If you engaged in an activity irrequentity (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a teacher's home.  Smoked organizes.  Drank beer.  Drank beer.  Drank organizes.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice after class.  Voted in a student election.  Socialized with someone of another racialisethnic group.  Came late to class.	at. If you y, mark y, one or mark y, one or mark w.	Important to you was esthe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b To be able to get a better. To gain a general education of ide. There was nothing better to make me a more cultuperson.  To be able to make more mit to learn more about thing that interest me.  To prepare myself for gracing professional school. A mentor/role model encouraged me to go. To get training for a specioaser.  To find my purpose in life.	ge CON CON CON CONTROL
(room, board, tuttion cover from each of the below? (Mark one a sach possible source (parelatives, spouse, each possible sources (sach po	n, and fees) do yo the sources listed inswer for it it irents sitc.) sitcs from ther income) is repaid as, military paid  stimate of your pa consider income fi as. (Mark one) 580,000 575,000 5100,000 5150,000 5200,000 5250,000	arents' total rom all -59,999 -99,999 0.149,999 0.0 or more	ones you did during the past veengaged in an activity irrequentity. If you engaged in an activity more times, but not frequently, (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Vias a guest in a teacher's home.  Smoked organeties.  Drank beer.  Drank beer.  Drank beer.  Drank beer.  Pett depressed.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice after class.  Voted in a student election.  Socialized with someone of another racial/ethnic group.  Came late to class.  Used the internet for research.	ar. If you y, mark y, one or mark of the state of the sta	Important to you was eithe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b To be able to get a better. To gain a general education of idea and appreciation of idea There was nothing better to make me a more cultuperson.  To be able to make more material to tearn more about thing that interest me.  To prepare myself for gracion professional school. A mentor/role model encouraged me to go.  To get training for a specioareer.  To find my purpose in life.  30. How would you charact political views? (Mark of political views?)	ge. CON
(room, board, tuttion cover from each of the below? (Mark sine a sect) possible source (paretatives, spouse, etc.)  Family resources (paretatives, spouse, etc.)  My own resources (see work, work, sudy, of Aid which need not be (grants, scholarship funding, etc.).  Aid which must be religions, etc.)  Other than above.  23. What is your best elincome last year? Coources before taxe.  Class than \$10.00  \$10.000-14.999  \$25.000-24.999  \$25.000-29.999  \$30.000-39.999  \$30.000-39.999	n, and fees) do yo the sources listed inswer for it it irents sto.) serepaid serepai	arents' total rom all -59,999 0-149,999 0-149,999 0-149,999 0-179,99 0-179,99 0-179,99 0-179,99 0-179,99 0-179,99 0-179,99 0-1	ones you did during the past ve engaged in an activity irrequentity. If you engaged in an activity irrequentity (Occasionally). Mark (Not at a if you have not performed the activity during the past year. (Mark one for each item)  Attended a religious service.  Was bored in class.  Participated in organized demonstrations.  Tutored another student.  Studied with other students.  Was a guest in a teacher's home.  Smoked organizes.  Drank beer.  Drank beer.  Drank organizes.  Performed volunteer work.  Played a musical instrument.  Asked a teacher for advice after class.  Voted in a student election.  Socialized with someone of another racialisethnic group.  Came late to class.	ar. If you y, mark y, one or mark of the state of the sta	Important to you was esthe following reasons? (Mark one answer for ear possible reason)  My parents wanted me to I could not find a job. Wanted to get away from b To be able to get a better. To gain a general education of ide. There was nothing better to make me a more cultuperson.  To be able to make more mit to learn more about thing that interest me.  To prepare myself for gracing professional school. A mentor/role model encouraged me to go. To get training for a specioaser.  To find my purpose in life.	ge. CON

Church of Christ 92 % 9	
Eastern Orthodox CP 20 9	···
Spiscopalian	
Hindia	3
Islamic	
Jewish	io [
LDS (Mormon) Si 3. 9	
Luneran	
Metanist © 20	(C)
Presbyletan	ė l
Current State	
Roman Catholic 900	
Seventh Day Adventist	W.
Unitariar/Universalist, ST	E.
United Church of Christ Congregational. (2) (2) 8	
Other Christian (1) II al	8
Other Religion	
None	
i. Are you: (Mark <u>all</u> that apply)	
Write Caucasien	**
African American/Black	Alley mon <sup>2</sup>
American Indian/Alaska Native	
Asian American/Asian	S [
Nativo Hawakan Pacto Islandar	^^. '**
Mexican American Chicano	···.
Puedo Ricen	
	2 1

Used a personal computer	4000
Discussed religion	000
Discussed politics:	
In class	<b>#</b>
With friends	(10 (10 15)
	###
Worked on a local state, or national political campaign.	
27. Dist your high school require service for graduation?	e community
28. What is the highest level of education obtained by your (Mark goe in each column)	
Gramman school or less	
Some high school	
High school graduate	
Postsecondary school other than college	
Same college	and the state of
College degrae	
College degrae	00
College degree Serve graduate school Graduate degree	

·2.

. Rate yourself on each traits as compared will person your age. We the most accurate estimate of how you see yourself. (Mark one in each row)	th the		
Apademic abidly			
Artistic ability		F. (2)	
Consputer skills	-24	(C)	and the
Cooperativeness		i Öd	
Creativity		. E	
Drive to achieve	(	: <u></u>	30
Emotoral health		-0	
Leadership ability			
Mathematical ability	100 mg		
Physical health	0	:	
Public speaking addity.	OC	1	50
Resignasments	00		
Sel-corhònice intelectual		Zw.	
Self-confidence (social)			
Sed-understanding	-00		
Spenicality			
Linderstanding of others		10	
Writing ability	ŌĈ		

ं Far right

Work <u>only three</u> responses, <u>one</u> in each column.	33. Mark <u>one</u> in each row:	① Dangres Strongly ① Disagres Somewhat ② Agree Somewhat
W Your možeri's occupation	reaction of the second of the	C Agrei Smorghy
Your talents occupation		ghts of criminals
	Aportion should be legal	<b></b>
NOTE: If your father or mother		
s deceased, please indicale	Marijuana should be liegalized	<b>3000</b>
his or her last occupation.	It is important to have laws prohibiting homosexu	al relationatrips
Vocambeto scusy	Racial discrimination is no longer a major problem	nin Arresica (IVI)
tator or emortainer	<ul> <li>Reassteally, an individual can do little to bring all</li> </ul>	nuc changes in our escenty
krohitect or urban planner		is than they do now
<b>vig</b>		wys. Eddi
Austress (clarical)	Same-sex couples should have the right to legal	marital status
Šijarkas kuncijum	Afternative action in callege admissions should be	e abstished
jmanagement, administrator) . S. C. C.	The autorities of exercised women are best confine	d to the borne and family
Susmess owner or proprietor 30.00	Federal military spending about be increased	
Submess salesparson or buyer 💢 🛴	If two necola really like each other it's all hold for	r flygeri \$1 isave sen everi if
Dergy (minister, priest)	they we known each other for dray a very Mort	line
Dergy (other maginus)	The tederal government should do more to contri	d the sale of handgors
Direction psychologist		
College administrator/staff 2 5		ontrol environmental poliution $\ldots \otimes \otimes \otimes \otimes$
Maje hadwi	and the second s	erybody's medical costs 🗊 🏵 🗇 🖰
Compuler programmer or analyst . 🗇 🗇	\$	y
Conservational or to ester		ess to public education
Jentist (natusing orthodoritist)	Through hard work, everylondy can succeed in A	merican society
Septem or multiformal		
rement	· · · · · · · · · · · · · · · · · · ·	
armen or resection	\$	
Foreign service worker (including diplomat)	34. During your last year in high school, how much time did you spend during a typical	36. Delow are some reasons that might have influenced your decision to affect this particular college. How important was each reason in your decision to come here?   Mark are answer for each possible reason;
Homemaker (full-time).		attend this particular college.
nauta decorator (including designer). 🗇 🗇	activities?	How important was each reason in your decision to come here?
ao technician or hygianist 🥽 🗇	Hours pot week:	attend this particular college. How important was each reason in your decision to come here? [Mark are answer for each possible reason]
aw entorcement officer	Harrist Married	
awyer (attorney) or judge 🖫 🖫	B <b>Studyinghomerick</b> DOCCOOD	My relatives wanted me to come here . (1) (3)
Military nervice (career) 🗇 🗇	Y Scouling with twenty 000000000	My teacher advised the 3. 35 8
Musician (performer, composer) . 🗇 🗇		This college has a very good
New		academic reputation
<b>description</b>		This college has a good reputation
Philemacks (C. C.)		for its social activities SSS
Physican	i www.cop. ocopooc	swas offered financial assistance . (1) (2) (3)
Policymaker/Government	OCCCOCO	The cost of attending this college (5) (5) (6)
School counselor 👙 🕄		High school counselor advised me 3 40 @
School principal or superintendent 😗 🖫		Private college courselor acrised me. 3.25
Scientific researcher	*	is warted to five reservoire
Serial malting or recognition worker		Not offered act by first choice

Therapist liphysical, occupational, spench)	Playing visitor	Trias cotege is graduates gain admission to top graduate!
Teacher or administrator pelementary)	computer garmes	professional actions
Teacher or administrator (secondary).	35. Do you have any concern about your ability	i was attracted by the religious affinishos/orientation of the cyllege (100) (200)
Veneraliza	to linance your college education?	warried to go to a school about the size of this college
Skilled trades	None (Lam confident that I will have sufficient function.	Rankings in national magazines (2006) — information from a website (2008) —
Semi-abilitied worker	Some (but ) probably will have enough funds:  Major (not sure ) will have enough funds by complete college)	A visit to the campus
Center:	Constrain Constrain	

# 

<ol> <li>Below is a list of different und fields grouped into general ca one oval to indicate your prob</li> </ol>	legories. Mark only	Please indicate the importance to you personally of each of the following:     (Mark goe for each titre)     Becoming accomplished in one of the	Not Important Somewhat Important Yery Inportant Essential
ARTS AND HUMANITIES	PHYSICAL SCIENCE		.,,,
Art, fine and applied	Astronomy	Seconing an authority in my field	
English (language and laterature)	Amospheric Science (incl. Meteorology)	Obtaining recognition from my colleagues is contributions to my special field	# 
History	Chemistry	influencing the political structure	
Journalism (2)	Earth Science	influencing succeif values	
Language and Literature	Varine Science (incl.	Raining a tamity	
(axcept English)	Dossnog sphy)	Having administrative responsibility for the s	work of others 🖫 🗓
Music	Mathematics	Seing very well off financially	
Philosophy	Physics	Helping others who are in difficulty	
Speach	Statestics	Making a theoretical contribution to science	
Theater or Drama	Other Physical Science	Writing original works (poems, novels, short	
Theology or Religion	PROFESSIONAL	Creating artistic work (painting, sculpture, d	
Other Arts and Humanides	Arrayamus or Urban	Becaming supposed in a business of my o	
BIOLOGICAL SCIENCE	Parmy	Becoming involved in programs to clean up :	tre environment 🗓 🗓
Biology (peneral)	Family & Consumer Sciences	Developing a meaningful philosophy of life	
Blochemistry or	Health Technology (med-	Participating in a community action program	ı
Supreyous 1	cal dental laboratory	Helping to promote racial understanding	
Bolany	Literary or Archival Science	Keeping up to date with political affairs	
Environmental Science	Magicine Genostry	Seconing a community leader	
Marine (Life) Science	Veternary Medicina	integrating spirituality into my life.	
Microbiology of	Nasing	improving my understanding of other counts	nes and cultures 🕮 🗓
Sactor except	Pharmacy	39. What is your best guess as to	14o Chance
Zxxiogy	Therapy (occupational)	the chances that you will:	Very Little Chance * Some Chance *
Other Biological Science	physical speechi	(Mark <u>ono</u> for each illem)	Very Good Chance
BUSINESS	Other Professional	Charge major text?	ÓÍ
Accounting	SOCIAL SCIENCE	Change career chocks?	
Busness Admin (general)	Arthropology	Participate in student government?	
Ference	Etonomos	Get a job to help pay for college expenses."	! E
International Business	Etrac Statios	Work hat time white attending college?	30
Marketing ®	Geography	Join a social haternity or sorbity?	
Management	Pulica Science (port	Play varsity/intercollegiate attentics?	
Secretarial Studies	interrigitorial relations).	Make at least a "6" average"	Œ
Other Business	Psychology	Participate in student protests or demonstra	abans?
EDUCATION	Social Work	Transfer to another college before graduati	ng? IJ
Business Education	Swawiogy	Be satisfied with your college?	🛣 🤅
Elementary Education - G	Wanter's Studens	Participate in volunteer or community servi-	oa work?
Music or Art Education	Other Social Science	Seex personal counseling?	
	TECHNICAL	Communicate regularly with your professor	
Physical Education or Recreation	Building Trades		
Secondary Education	*	Participate in student dubacy aups?	
Special Education	Data Processing or Computer Programming		
SUPERIOR SUPERIOR SUPERIOR STATE STA	mencely contact is a profession and all the	Participate in a study abroad program?	

ENGINEERING	Heatrance	***	and the second s	1077 WM 1
Aeronal/lical or	Mechanics		r Education Research Institute (I or should your college request th	
Astronautical Eng	Other Technical		ERI maintains sirici standards o	
Civil Engineering	OTHER FIELDS	would require your co	dege to sign a pledge of conlide	ntiality. — Yes — No. 1
Chemical Engineering ③	Agriculture			
Computer Engineering	Communications 5		swided for questions specifically o cation Ressarch Institute. If your c	
Electrical of Electronic	Computer Science		are fully the supplemental directio	
■ Engravero 1	Forestry 2	41. 11 11 11 11 11 11	48. 2 3 3 3 5 5 7	55. CO CO
■ Industrial Engineering	Knesiology	<b>42.</b> D D D D E D	<b>4.</b> FF 6 7	<b>56.</b>
■ Mechanical Engineering	Lax Entercement	43. @ @ @ 4 4	<b>50</b> . (30. (20. (20. (20. (20. (20. (20. (20. (2	<b>57.</b> (3.40 %) (3.40
■ Coner Engineering	Mittary Science	44. (B.3) (C.T.)	<b>51.</b> 3 (3) (3) (3) (3)	58. I I I O O O
	Other Field	45. 3 30 G G G	52. (3. (2. (2. (2. (2. (2. (2. (2. (2. (2. (2	59. 0 10 10 10 10
	Undecided	46. G D D D D D	<b>53.</b> (2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2	<b>60.</b> (4) (4) (5) (5)
<u> </u>	managan	47. I COOKE	<b>4</b> .44466	61, 400000
		G Prepared by the Higher Educ of Children's Lon Anglies. C	allyn Disaach in altain, University actorna 18096-1521	THANK YOU!
		. 4 -	Sima Macogra	mar Carp. (SSS) (1967) (1967)

. . .

#### APPENDIX C

#### Dear Students:

I am a graduate student in the Department of Counseling and Student Development. I am conducting research for my master's degree in College Student Affairs. This study is being conducted to develop a comprehensive understanding of the expectations, perceptions and actual experiences of African American undergraduates at the university.

Data collection will be obtained by survey methodology, and the information obtained from this study will be kept COMPLETELY CONFIDENTIAL and anonymous. There are no foreseeable risks or benefits to the participant. You will not be asked to give your name or any information that would identify you as an individual. Individuals are in no way required to participate in the study. If individuals agree to participate, they may at any time discontinue the survey or choose to not answer any questions at any time. This survey must be completed in a single setting; you may not begin the survey and then finish on a later date. The survey will likely take 20-25 minutes to complete.

In agreement to participate in this study, each student will have the opportunity to enter himself or herself into the random drawing for one of two \$25 VISA GIFT CARDS.

BY CLICKING ON THE CONSENT BUTTON BELOW, YOU GIVE YOUR CONSENT.

#### CONSENT

After clicking on the CONSENT button, you will be directed to the survey.

Thank you for your anticipated participation in this study. I greatly appreciate your cooperation. Sincerely,

Christina L. Rainer Principal Researcher

Christina L. Rainer Graduate Assistant Advisor Gateway Program Eastern Illinois University (217)581-0742 clrainer@eiu.edu